INSTITUTO TECNOLÓGICO Y DE ESTUDIOS SUPERIORES DE OCCIDENTE

Reconocimiento de validez oficial de estudios de nivel superior según acuerdo secretarial 15018, publicado en el Diario Oficial de la Federación el 29 de noviembre de 1976.

Departamento de Electrónica, Sistemas e Informática

Doctorado en Ciencias de la Ingeniería



MODELO HÍBRIDO DE SIMULACIÓN DE SERVICIOS PARA LA IMPLEMENTACIÓN DE ECONOMÍA CIRCULAR

Tesis que para obtener el grado de Doctor en Ciencias de la Ingeniería presenta: Edna Leodegaria Guevara Rivera

Director de tesis: Dr. Roberto Osorno Hinojosa Co-director de tesis: Dr. Víctor Hugo Zaldívar Carrillo

Tlaquepaque, Jalisco. Noviembre de 2022

TITULO:	Modelo Híbrido de Simulación de Servicios para la Imple-
	mentación de Economía Circular
AUTOR:	Edna Leodegaria Guevara Rivera
	Licenciada en Ciencias de la Electrónica (BUAP, México)
	Maestra en Ciencias de la Electrónica con Opción en Automati-
	zación (BUAP, México)
	Maestra en Sistemas de Calidad y Productividad (ITESM, Méx-
	ico)
DIRECTOR DE TESIS:	Roberto Osorno Hinojosa
	Departamento de Electrónica, Sistemas e Informática, ITESO
	Ingenierio en Sistemas Computacionales (ITESO, México)
	Maestro en Informática Aplicada (ITESO, México)
	Doctor en Economía y Empresa (Universidad Ramon Llull, Es-
	paña)

NÚMERO DE PÁGINAS: XXI, 141

ITESO – The Jesuit University of Guadalajara

Department of Electronics, Systems and Informatics DOCTORAL PROGRAM IN ENGINEERING SCIENCES



HYBRID SERVICE SIMULATION MODEL FOR CIRCULAR ECONOMY IMPLEMENTATION

Thesis to obtain the degree of DOCTOR IN ENGINEERING SCIENCES Presents: Edna Leodegaria Guevara Rivera

Thesis Director: Dr. Roberto Osorno Hinojosa Thesis Co-director: Dr. Víctor Hugo Zaldívar Carrillo

Tlaquepaque, Jalisco, Mexico

November 2022

TITLE:	Hybrid Service Simulation Model for Circular Economy Im-
	plementation
AUTHOR:	Edna Leodegaria Guevara Rivera
	Bachelor's degree in Electronics Sciences (BUAP, México)
	Master's degree in Automation (BUAP, México)
	Master's degree in Quality and Productivity Systems (ITESM,
	México)
THESIS DIRECTOR:	Roberto Osorno Hinojosa
	Departament of Electronics, Systems, and Informatics, ITESO
	Bachelor's degree in Computer System Engineering (ITESO, Mex-
	ico)
	Master's degree in Applied Informatics (ITESO, Mexico)
	Ph.D degree in Economy and Enterprise (University Ramon Llull,
	Spain)

NUMBER OF PAGES: XXI, 141

Acknowledgements

Throughout the writing of this dissertation I have received a great deal of support and assistance. I would first like to acknowledge the authorities of ITESO and CONACyT for the financial support through an assistantship granted for the postgraduate accomplishment.

I would like to thank my supervisors, Dr. Roberto Osorno Hinojosa and Dr. Victor Hugo Zaldívar Carrillo, whose expertise was invaluable in formulating this methodology. Their insightful feedback pushed me to sharpen my thinking and brought my work to a higher level.

I would like to acknowledge my colleagues from ITESO university. I would particularly like to single out Dr. Humberto Pérez Ortiz, I want to thank you for your patient support and for all of the opportunities I was given to further my research.

I would also like to thank my tutors, Dr. José Ernesto Rayas Sánchez, Dr. Zabdiel Brito Brito, and Dr. José Francisco Cervantes Álvarez for giving me the opportunity to be part of the postgraduate program and for their valuable guidance throughout my studies. You provided me with the tools that I needed to choose the right direction and successfully complete my dissertation.

In addition, I would like to thank my brothers José Luis and Esau for their support and sympathetic ear; and to my friends Maru Guerrero, and Ale Pacheco, who always encouraged me to continue. You are always there for me.

I could not have completed this dissertation without the understanding and love of my husband, Leonardo Sandoval González, who encouraged and supported me from the beginning till the end of this adventure.

Finally, I specially dedicate this work to my children Leo and Caro, who are my engine to continue growing professionally and emotionally.

Resumen

Los principios de la economía circular (EC) se han creado en respuesta al agotamiento de los recursos naturales como un conjunto de pautas para eliminar el modelo lineal de toma-uso-disposición de consumo de productos. Las consecuencias de pasar de una cadena de suministro lineal a una circular son difíciles de visualizar a largo plazo. Por tanto, implementar una herramienta de simulación de economía circular en procesos lineales de pequeñas y medianas empresas (Pymes) es fundamental para probar políticas antes de implementarlas en el mundo real. Este estudio tuvo como objetivo evaluar la lógica dominante de servicio, los servicios ecosistémicos, la dinámica de sistemas y el modelado basado en agentes para diseñar una metodología para un modelo de simulación implementado en dos estudios de caso: un banco de alimentos y una fábrica de confitería. En ambos casos se hicieron visitas y entrevistas con las partes interesadas para evaluar el modelo de simulación durante la fase de desarrollo. El prototipo de indicador de economía circular (CEIP), cuya puntuación fue del 52 % (calificado como un producto "bueno"), se utilizó como medida de madurez circular de la fábrica de confitería. Los simuladores para cada caso de estudio se ejecutaron en el software Netlogo implementando un análisis escenarios basado en políticas de CE. Se utilizaron diversas variables en estos análisis relacionadas con los costos del proceso y con la cantidad de producto deshechado y reciclado. El principal aporte de este trabajo es la metodología implementada en dos casos de estudio reales en México en el que se diseñó dos modelos de simulación para evaluar estrategias de economía circular en escenarios futuros. Además, en el caso de la fábrica de confitería el simulador permitió a los interesados comprender el funcionamiento del proceso de reciclaje y visualizar todas las variables involucradas en el sistema.

Summary

The circular economy (CE) principles have been created in response to the depletion of natural resources as a set of guidelines to eliminate the linear take-use-disposition model of product consumption. The consequences of moving from a linear supply chain to a circular one are difficult to visualize in the long term. Therefore, in some cases, implementing a circular economy simulation tool in linear processes of small and medium-sized enterprises (SMEs) is essential to test policies before implementing them in the real world. This study aimed to evaluate the dominant service logic, ecosystem services, system dynamics, and agent-based modeling to design a methodology for a simulation model implemented in two life cycle case studies: a food bank and a confectionery factory. In both cases, visits and interviews were made with interested parties to evaluate the simulation model during the development phase. The circular economy indicator prototype (CEIP), whose score was 52% (rated as a "good" product), was used as the circular maturity measure of the confectionery factory. We used NetLogo software to execute the simulation models for each case study, implementing a scenario analysis based on CE policies. Various variables were used in these analyses related to the process's costs and the amount of discarded and recycled products. The main contribution of this work is the methodology implemented in two real case studies in Mexico, in which we designed two simulation models to evaluate circular economy strategies in future scenarios. In addition, in the case of the confectionery factory, the simulator allowed interested parties to understand the operation of the recycling process and to visualize all the variables involved in the system.

Contenido

Agradecimientos	V
ResumenVI	Π
Summary E	X
Contenido	X
ContentsXI	Π
Lista de FigurasXVI	Π
Lista de TablasXX	X
Introdución	1
1. Revisión de la Literatura	5
2. Caso de Estudio y Propuesta de la Metodología 1	7
3. Jalisco sin Hambre: Banco de Alimentos de Tepatitlán 2	27
4. Modelación Basada en Agentes 4	1
5. Construyendo el Modelo de Simulación en Netlogo 5	;5
6. Validación del Modelo 6	i9
7. Segundo Caso de Estudio: Fábrica de Dulces	7

8. Implementación de la Metodología en la Fábrica de Caramelo	85
9. Mediciones de Economía Circular	
10.Análisis de Escenarios	101
11.Actualización de la Revisión de la Literatura	107
General Conclusions	119
Conclusiones Generales	121
Apéndice	
A. Lista de Reportes Internos de Ivestigación	
B. Lista de Publicaciones	
Bibliografía	127

Contents

Ac	knov	wledgements	V
Re	sum	enV	II
Su	mma	aryI	[X
Co	onten	iido	X
Co	onten	ntsX	II
Li	st of	FiguresXV	II
Li	st of	TablesX	X
In	trodu	uction	1
1.	Lite	erature Review	5
	1.1.	Circular Economy	5
	1.2.	Service-Dominant Logic	8
	1.3.	Ecosystem Services	11
	1.4.	System Dynamics and Agent-Based Modeling	13
		1.4.1 System Dynamics	14
		1.4.2 Agent-Based Modeling	15
	1.5.	Conclusions	16
2.	Cas	e Study and Methodology Proposal	17
	2.1.	Case Study: "Jalisco sin Hambre"	17

Contents

	2.2.	Obtain	ing the Methodology Proposal	19
		2.2.1	Select interacting parts of the value chain	22
		2.2.2	Analyze and choose the CE strategies for each part or entity	23
		2.2.3	Define the entities' attributes according to CE strategies	24
		2.2.4	Define the agent-based model for the selected entities	25
	2.3.	Conclu	sions	25
3.	"Jal	lisco si	n Hambre": Tepatitlán Food Bank	27
	3.1.	Food B	ank Process	28
	3.2.	Method	dology Steps	29
		3.2.1	Choose interacting parts of the value chain (entities)	30
		3.2.2	Analyze the CE Strategies for each part or entity	30
		3.2.3	Define the entities attributes according to the CE strategies	32
		3.2.4	Define the agent based model for the selected entities	33
		3.2.5	Define the dynamic behavior of the environment variables	34
		3.2.6	Map a Diagram connecting the entities or agents with the environment	34
		3.2.7	Build the simulation model in a software	35
		3.2.8	Define and execute performance tests	36
		3.2.9	Formulate and evaluate policies	36
	3.3.	Conclu	sions	38
4.	Age	nt-Bas	ed Modeling	41
	4.1.	Introdu	iction	41
	4.2.	The M.	AIA Meta-Model in the case study JSH	42
		4.2.1	Collective Structure	42
		4.2.2	Constitutional Structure	46
		4.2.3	Physical Structure	47
		4.2.4	Operational Structure	49
		4.2.5	Evaluative Structure	52
	4.3.	Conclu	isions	54

5.	Buil	lding t	he Simulation Model in Netlogo	55
	5.1.	Introdu	ction	55
	5.2.	ODD F	Protocol	55
	5.3.	Tepatit	lán Food Bank ODD Protocol	55
		5.3.1	Purpose	56
		5.3.2	Entities, State Variables, and Scales	56
		5.3.3	Process overview and scheduling	56
		5.3.4	Design concepts	57
		5.3.5	Initialization	60
		5.3.6	Input data	60
		5.3.7	Submodels	61
	5.4.	First Re	esults of Model Execution	65
	5.5.	Conclu	sions	66
6	Ма		idation	<i>(</i>)
0.				09
	6.1.	Introdu		69
	6.2.	Model	Evaluation	69
		6.2.1	Boundary adequacy test	69 70
		6.2.2	Structure assessment test	70
		6.2.3	Dimensional consistency	71
		6.2.4	Parameter assessment	71
	6.3.	Model	Validation	71
	6.4.	Conclu	sions	72
7.	Seco	ond Ca	se Study: Confectionery Factory	77
	7.1.	Introdu	ction	77
	7.2.	Food ir	ndustry	77
	7.3.	Circula	r economy analysis	78
	7.4.	Case St	tudy: Confectionery Factory	79
	7.5.	Step 1:	Select interacting parts of the value chain	81
	7.6.	Conclu	sions	82

8.	Met	hodology Implementation in the Confectionery Factory	85
	8.1.	Introduction	85
	8.2.	Step 2. Analyze the CE strategy for each part or entity	85
	8.3.	Step 3. Define the entities' attributes according to the CE strategies	86
	8.4.	Step 4. Define the agent-based model for selected entities	87
	8.5.	Step 5. Define the dynamic behavior or a mathematical model of the environment	
		variables	87
	8.6.	Step 6. Map a diagram connecting the entities or agents with the environment	87
	8.7.	Step 7. Build the simulation model in a software	88
	8.8.	Step 8. Define and execute performance tests	89
	8.9.	Step 9. Formulate and evaluate policies	90
	8.10.	Conclusions	92
9.	Circ	cular Economy Measurements	93
	9.1.	Introduction	93
	9.2.	CE measures	94
		9.2.1 Circularity index	94
		9.2.2 Break-even point	95
	9.3.	Case study: confectionery factory	96
		9.3.1 Circularity index	96
		9.3.2 Break-even point	97
	9.4.	Conclusions	98
10	.Scer	nario Analysis1	01
	10.1.	Introduction	101
	10.2.	Step 8: Performance Test Execution	102
	10.3.	Step 9: Policy Formulation and Evaluation	103
	10.4.	Model Validation	105
	10.5.	Conclusions	106
11	.Lite	rature Review Update1	107

11.1	. Introduction
11.2	2. Circular Economy
11.3	B. Service-dominant logic (S-D Logic)
11.4	Ecosystem Services
11.5	5. Agent-based modeling
11.6	5. System Dynamics
11.7	7. Hybrid Modeling
11.8	B. Conclusions
Gener	al Conclusions
Gener Conclu	al Conclusions
Gener Conclu Appen	al Conclusions
Gener Conch Appen A.	al Conclusions
Gener Conclu Appen A. B.	al Conclusions

List of Figures

1.1	Circular Economy proposal from Ellen MacArthur Foundation (figure taken from[33])	6
1.2	Circular economy for sustainable development in three dimensions: economic, en-	
	vironmental, and social. (figure taken from [70])	7
1.3	The narrative and process of S-D logic [134].	10
1.4	A service-dominant value creation (SVC) framework for ecosystem service offer-	
	ings in value co-creation within the socio-ecological system (figure taken from [86])	13
1.5	The modeling process of System Dynamics (figure taken from [120])	14
2.1	Circular economy principles proposed by Webster [138]	19
2.2	Circular economy adoption from its principles assigning the studied methodologies	
	and tools.	20
2.3	Circular economy resource flows through a value chain (taken from [67])	22
2.4	Methodology proposal for implementing a hybrid-service simulation model for CE	23
3.1	The fifteen regions attended by the Tepatitlán Food Bank in Jalisco.	27
3.2	Total food received at Tepatilán FB during 2017 (kg)	28
3.3	Tepatitlán food bank process, from the food donations until the food bag delivery	
	to families.	29
3.4	Circular value chain for biological cycles (obtained from [82])	30
3.5	Food waste hierarchy strategies (taken from [95])	31
3.6	Three layers meta-model. (taken from [43])	32
3.7	Causal diagram of the natural environment related to food production	34
3.8	Interaction among environment variables and food production, using stocks and flows.	35
3.9	Tepatitlán food bank operations using stocks and flows.	36
3.10	Tepatitlán food bank operations using stocks and flows	37

LIST OF FIGURES

3.11	Life Cycle Analysis elements for any process (taken from [8])	38
4.1	Physical structure represented as flows and stocks in a system dynamics model	49
4.2	Entity actions related to the flows in the physical structure.	51
4.3	Agent interaction with their physical structure.	52
5.1	System dynamics model for Tepatitlán food bank.	57
5.2	Causal diagram for Tepatitlán food bank	58
5.3	Agent interaction with their physical structure	60
5.4	Tepatitlán food bank simulation setup.	65
5.5	Tepatitlán food bank simulation execution.	66
6.1	Causal diagram for defining the model boundary of the Tepatitlan Food Bank	70
6.2	Data collected from Tepatitlán Food Bank	74
7.1	Value chain parts involved in the confectionery waste recycling process (adapted from [67])	82
8.1	Processes connected with I/O variables.	86
8.1 8.2	Processes connected with I/O variables.	86 88
8.18.28.3	Processes connected with I/O variables	86 88 89
8.18.28.38.4	Processes connected with I/O variables	86 88 89 90
 8.1 8.2 8.3 8.4 8.5 	Processes connected with I/O variables	86 88 89 90 90
 8.1 8.2 8.3 8.4 8.5 8.6 	Processes connected with I/O variables	86 88 89 90 90 91
 8.1 8.2 8.3 8.4 8.5 8.6 8.7 	Processes connected with I/O variables	 86 88 89 90 90 91 92
 8.1 8.2 8.3 8.4 8.5 8.6 8.7 9.1 	Processes connected with I/O variables	86 88 89 90 90 91 92 97
 8.1 8.2 8.3 8.4 8.5 8.6 8.7 9.1 9.2 	Processes connected with I/O variables	 86 88 89 90 90 91 92 97 99
 8.1 8.2 8.3 8.4 8.5 8.6 8.7 9.1 9.2 10.1 	Processes connected with I/O variables	86 88 90 90 91 92 97 97 99
 8.1 8.2 8.3 8.4 8.5 8.6 8.7 9.1 9.2 10.1 10.2 	Processes connected with I/O variables	 86 88 89 90 90 91 92 97 99 101 102
 8.1 8.2 8.3 8.4 8.5 8.6 8.7 9.1 9.2 10.1 10.2 10.3 	Processes connected with I/O variables.	 86 88 89 90 91 92 97 99 101 102 103
 8.1 8.2 8.3 8.4 8.5 8.6 8.7 9.1 9.2 10.1 10.2 10.3 10.4 	Processes connected with I/O variables. . Dynamic behavior of the confectionery case study. . Simulation model of the confectionery case study. . The behavior of the confectionery manufacturing and recycling production . The behavior of confectionery waste vs recycled glucose stocks. . The behavior of recycling production when this is sold to other industries. . The behavior of recycling production when waste from other industries is recycled. Circularity index in the caramel recycling process. . Break-even point in recycling caramel process. . Performance of primary variables in the hybrid simulation model. . Four scenarios are proposed from two policies. . Profit performance from the four scenarios related to the two policies: marketing	 86 88 89 90 91 92 97 99 101 102 103

11.1	Comparing the conceptualization of value and focal actors of S-D logic and related
	perspectives (taken from [132])
11.2	The broad applicability of S-D logic and existing diffusion vectors in diverse disci-
	plines. (taken from [132])
11.3	System Dynamics in the evolution of systems approach (taken from [117]) 113

List of Tables

1.1	Six limits and challenges for the Circular Economy concept [70]	9
1.2	The axioms of Service-Dominant Logic	11
1.3	Research frontiers of Service-Dominant Logic	12
1.4	System Dynamics vs Agent-Based Modeling simulation [116]	16
2.1	Literature related to the interceptions between methodologies	21
2.2	Steps of the Methodology proposal to implement a CE simulation model	24
3.1	Identification of the meta-model elements in JSH	33
4.1	Collective Structure of JSH agent model	43
4.2	Decision-Making Criterion	44
4.3	Constitutional Structure	47
4.4	Institutions	48
4.5	Physical Structure	48
4.6	Action Situations	50
4.7	Scope matrix of Evaluative Structure	53
5.1	Initialization of System Dynamics Model	61
5.2	Initialization of Agent-Based Model	62
6.1	Variables Dimension	73
6.2	Parameter Estimation	75
7.1	CEIP instrument applied to the confectionery factory	81
9.1	Calculus of Break-even point	98

10.1	Variables observed in the proposed scenarios.	104
11.1	The Business Ecosystem Modeling With The Integration of System Modeling and	
	Ecosystem Theory	114
11.2	The Business Ecosystem Modeling With The Integration of System Modeling and	
	Ecosystem Theory (cont.)	115
11.3	Similarities between Hybrid Service Simulation Model (HSSM) and Business Ecosys-	
	tem Modeling framework	117

Introduction

The current and traditional linear production system is unsustainable due to the traditional oneway (linear) throughput flow of materials and energy between nature and the human economy. In response to this global sustainable issue, a circular economy aims to redefine growth by proposing a cyclical flow of materials and energy [47, 70]. However, the outcome when transitioning from a linear to a circular supply chain system is not readily visible in the long term. For this reason, stakeholders fear investing in sustainable policies, and it is not easy to convince them [70]. Following this premise, there are academic contributions related to developing frameworks and simulators that support stakeholders during the decision-making process when implementing CE strategies within their organization. For example, [98] developed a framework to implement business model innovations related to CE. Likewise, [73] combined agent-based (ABM) and discrete-event simulation (DES) models to develop a tool where the product's design and production were evaluated in different scenarios according to its carbon footprint. Furthermore, DES and system dynamics (SD) theory are used to develop a simulation model that helps identify a correct CE strategy when upgrading, refurbishing, remanufacturing, or recycling the desired product [19]. In this sense, [56] designed a hybrid services simulation model (HSSM), a nine-step methodology for designing a simulation tool to prove several circular policies before implementation. This methodology is based on four perspectives: service-dominant logic [133], ecosystem services [86], agent-based modeling [140], and system dynamics theory [46].

The EMAF (Ellen MacArthur Foundation) classified the flow of materials through the supply chain into two different cycles: biological and technical [33]. EMAF published a report on food systems and their performance in big cities regarding biological cycles. This paper explains why the current feeding systems are no longer sustainable. Several causes can be attributed to the substantial quantities of food waste generated by large cities, including food production processes that are harmful to humans and the long distances between food production sites and cities that increase the food surplus and packing. Addressing these problems, EMAF proposes three objectives that cities can accomplish to catalyze a circular economy for food: (1) source food is grown regeneratively

INTRODUCTION

and locally where appropriate, (2) make the most of food, and (3) design and produce healthier food products [32].

Moreover, the food industry may not be producing enough nutritious food for the growing global population, leading to alterations in the overall nutritional needs [128]. Besides, water scarcity, energy use, and land availability need to be addressed to transform a food system into a healthy, nutritious, and environmentally sustainable practice [42, 88, 130].

The methodology proposed in [56] was implemented in a food bank as a case study in [55] regarding the food sector. We concluded from this implementation that food processes are complex and different from regular products even though it behaves like a value chain in this industry. This complexity varies depending on the food production stage, from farming and collection to the final disposal. The confectionery sector is not the exception. The challenges faced by food processes are essential in the connection between nutrition, health, and environmental issues [128]. On this basis, the increasing consumption of confectionery products is pressuring global supply chains. For example, cocoa and palm oil are only grown in certain parts of the world [88], and in 2018, Mexico was the second place in confectionery consumption in Latin America and the sixth place in the world, with an estimated consumption per capita of 4.5 kg per year [31].

Since confectionery products are diverse, the entire supply chain is complex, like raw material supply, specialized equipment acquisition, tasty and nutritious recipe design, product packing, commercialization strategies, and the final disposal of confectionery products. A system-view approach across the confectionery supply chain is required from raw material use to the final disposal, considering a cradle to grave assessment to attend to the transition towards environmental sustainability [88]. There is a significant growth in confectionery waste valorization in the European Union, for example, the production of bioethanol and biogas as bioenergy products and bio-based and biodegradable polymers [58].

Therefore, this study aims to assess service-dominant logic, ecosystem services, system dynamics, and agent-based modeling to design a methodology for a simulation model implemented in a Mexican small and medium-sized enterprise [38], particularly in a confectionery factory with a value chain configuration within its business model. The obtained simulator will allow stakeholders to analyze and decide on the CE strategies to implement in their business process.

This doctoral dissertation is organized as follows:

2

INTRODUCTION

Chapter 1 presents an overview of the circular economy and the four perspectives used in the proposed methodology design.

Chapter 2 explains how the Methodology was obtained from the four perspectives.

Chapter 3 introduces the problem of the first case study, a food bank located in Jalisco, México.

Chapter 4 develops the agent-based model for the food bank case study.

Chapter 5 explains the implementation of the simulation model in Netlogo software.

Chapter 6 presents some validation proposals for the simulation model obtained.

Chapter 7 explains the context of the second case study, a confectionery factory.

Chapter 8 presents the methodology implementation of the second case study.

Chapter 9 describes the different indicators to measure circular economy implementation in circular supply chains.

Chapter 10 analyzes the proposed scenarios for the confectionery case study.

Chapter 11 presents an update of the recent literature concerning the circular economy and the four perspectives.

Finally, this thesis includes two appendices. Appendix A shows the reference list of the eleven internal research reports presented during the doctoral studies, and Appendix B shows the list of conference and journal papers published.

INTRODUCTION

1. Literature Review

1.1. Circular Economy

Circular economy (CE) is a concept recently used in the European Union as an alternative for producing products and services [70]. Instead of using the traditional industrial scheme "make-usedispose," CE proposes to imitate nature's way of using the resources and renewing the waste produced by nature's process. Whereas industrial processes commonly use energy from non-renewable resources, nature only uses solar energy to accomplish its processes. The waste generated through these processes is reintegrated into the soil, and when the living beings achieve their end of life, they serve as food to other microorganisms. According to Webster [138], the circular economy rests on five principles:

- a) Design out waste: "Waste does not exist when the biological and technical components (or 'materials') of a product are designed to fit within a biological or technical materials cycle, designed for disassembly and re-purposing."
- b) Build resilience through diversity: "Diverse systems with many connections and scales are more resilient of external shocks than systems built simply for efficiency – throughput maximization driven to the extreme results in fragility."
- c) Work towards using energy from renewable sources: "Systems should ultimately aim to run on renewable energy – enabled by the reduced threshold energy levels required by a restorative, circular economy."
- d) Think in systems: "The ability to understand how parts influence one another within a whole, and the relationship of the whole to the parts, is crucial."
- e) Think in cascades: "For biological materials, value creation lies in the opportunity to extract additional value from products and materials by cascading them through other applications."

These proposed principles are similar from those three that the Ellen MacArthur Foundation (EMAF) explains [36]:

- a) "Preserve and enhance natural capital. By controlling finite stocks and balancing renewable resource flows."
- b) "Optimize resource yields. By circulating products, components, and materials at the highest utility at all times in both technical, biological cycles."
- c) "Foster system effectiveness. By revealing and designing out negative externalities."

These three principles are better explained in Fig. 1.1, where the circular material flows proposed by CE are shown. Furthermore, Fig. 1 shows the different levels where these principles are applied. Both proposals, those from Webster [138] and EMAF [36], are similar. This chapter, makes several references to these principles to explain their impact.



Figure 1.1: Circular Economy proposal from Ellen MacArthur Foundation (figure taken from[33])

The main objective of the EMAF is to "create evidence-based original research on the benefits of a CE, and how it can contribute to solving global challenges like climate change and biodiversity loss" [33]. The EMAF is formed by European stakeholders who have contributed to implement the concept of CE in some industries. On their web page, it is possible to find some successful

study cases [33]. On the other hand, it is necessary to measure some indicators to determine the actual processes' ecologic impact. EMAF proposes the Material Circulating Indicator (MCI) in industrial processes. This indicator is calculated from other measurements of recycling products and refurbishing a specific industrial process [80]. The European Union established a standard that contains the ecological impact of some materials during their fabrication and consumption: "the lifecycle assessment (LCA) is a widely applied methodology in the context of environmental analysis to support cleaner production and greener supply chains" [48]. Besides, [62] proposes several metrics for evaluating CE in several dimensions, like economic, social, environmental, and technical. In future work, these metrics might help evaluate the case study performance according to CE.



Figure 1.2: Circular economy for sustainable development in three dimensions: economic, environmental, and social. (figure taken from [70])

CHAPTER 1. LITERATURE REVIEW

Beyond these concepts, the implementation of CE on a physical system depends on several factors, and it is necessary to consider the limitations of CE, as Korhonen recently did [70]. In his article, he suggests six limitations according to the impact of the CE implementation in three aspects: environmental, economic, and social (see Fig. 1.2). The limitations summarized in Table 1.1 refer to real problems found when the systems are converted to CE. We focused on the fifth limitation: "Limits of governance and management" (intra-organizational vs. inter-organizational strategies and management), where the author discusses the willingness of the participants or actors for changing to a circular working mode with the critical question: "How can an individual firm convince its stakeholders, customers, and authorities that its strategy of waste maximization is beneficial for the environment and sustainability?" [70]. In the next sections we will present the importance of a simulation model for explaining and convincing to decision makers for adopting CE in their processes.

1.2. Service-Dominant Logic

We explored another concept concerning service ecosystems. Vargo and Lusch in [133] stated that marketing or economic activity is best understood in terms of service-for-service exchange rather than an exchange in terms of goods-for-goods or goods-for-money. In other words, the source of value in a service ecosystem consists of the activities emanating from specialized knowledge and the abilities that people do for themselves and others (i.e., service, applied abilities), not the goods. Value is co-created by several actors and subsequently delivered (see Fig. 1.3).

Vargo and Lusch [134] identified eleven foundational premises, and five have become axioms (see Table 1.2). These axioms serve as a foundation of Service-Dominant (S-D) logic. Furthermore, these authors have publicized other inputs to this research line. One of their most recent articles exposes future research concerning the S-D logic from different perspectives [135]. In this article, the need to find a meta-theory that explains the use of S-D logic is assessed. Among other perspectives, the system and ecosystem theory is explained from its roots, which are in "natural" science and attributed to Tansley as "the basic unit of nature" [125]. In SDL, the term ecosystem is used in the business environment as "networked constellations of firms, often centered on a central actor" [136].

TABLE 1.1. SIX LIMITS AND CHALLENGES FOR THE CIRCULAR ECONOMY CONCEPT [KORHONEN]

Limit	Description
Thermodynamic limits	Cyclical systems consume resources and create wastes and emissions
System boundary limits	Spatial: problems are shifted along with the product life cycle. Temporal: Short-term non- renewables use can build long-term renewable infrastructure
Limits posed by the physi- cal scale of the economy	The rebound effect, Jevon's paradox, the boomerang effect
Limits posed by path- dependency, and lock-in	First technologies retain their market position despite in-efficiency
Limits of governance and management	Intra-organizational and intra-sectoral manage- ment of inter-organizational and inter-sectoral physical material and energy flows.
Limits of social and cultural definitions	The concept of waste strongly influences its han- dling, management, and utilization. The con- cept is culturally and socially constructed, and the concept of waste is always constructed in a particular cultural, social, and temporal context that is dynamic and changing.

Vargo and Lusch propose several aspects for studying their S-D logic impact on the future. One of them is related to dynamic strategy development and implementation. There is a need to implement theories of strategy development "to develop ongoing, dynamic, cooperative relationships that enable access to and integration of resources resulting in new resources" [135]. Emphasizing: (1) considering the ecosystem diversity where the organization is located, and (2) the collaboration between other actors and beating the competition. This issue is summarized in the following question: "How can strategic planning and implementation be co-created with multiple stakeholders, and what is the impact of these co-creation processes on the firm and its stakeholders?" [135]. Another aspect is complexity economics, which studies the economic and social actors under more

CHAPTER 1. LITERATURE REVIEW



Figure 1.3: The narrative and process of S-D logic [134].

realistic assumptions, such as non-linearity, actor created rules that can become law-like through institutionalization within a network of other actors (also named ecosystem). Complexity economics uses computational economics, which integrates computer science and learning with economics. The computational economic tools can be agent-based modeling, cellular automata, and genetic algorithms, among others, to model and understand complex service ecosystems and the broader economy comprising them. The question proposed by Vargo and Lusch in this sense is: "How can concepts from complexity economics be used to develop a general model of a complex service ecosystem that could then be used to further research on markets and the economy?" [135].

The third theme we consider essential for this research is environmental sustainability. Vargo and Lusch establish that S-D logic, focusing on service ecosystem viability and resiliency, can be used as an informative and robust framework for environmental sustainability. The salient research question concerning this theme is: "How can S-D logic and ecosystem service(s) be used to advance environmental sustainability?" [135]. We have considered these aspects the most critical issues

Number	Axioms
1	Service is the fundamental basis of exchange
2	Value is co-created by multiple actors, always in- cluding the beneficiary
3	All social and economic actors are resource inte- grators
4	Value is always uniquely phenomenologically de- termined by the beneficiary
5	Value co-creation is coordinated through actor- generated institutions and institutional arrange- ments

TABLE 1.2. THE AXIOMS OF SERVICE-DOMINANT LOGIC

that the simulator software will impact. However, depending on the research inclination, the results could answer other questions that Vargo and Lusch propose [135]. These questions are summarized in Table 1.3.

1.3. Ecosystem Services

Regarding environmental sustainability, Matthies *et al.* [86] compare the ecosystem services (ES) with the service system. ES refers to the benefits humans obtain from natural ecosystems for their well-being. In a service system or ecosystem, the word 'service' refers to the process of doing something beneficial for and in conjunction with some entity [136]. Thus, the ES approach is potentially an extension of service sciences. The creation of value by one service system is a nonlinear interactive and dynamic process (see Fig. 1.3). and results are a potential value that can be utilized, missed, or destroyed by various other actors, processes, and resources that are part of a service systems' value network (i. e. a forest). From this definition, Matthies indicates the role of the natural ecosystem as a service system providing offerings with potential value and proposes a service-dominant value creation (SVC) framework for ecosystem service offerings in value co-creation within a socio-ecological system. In Fig. 1.4, the natural ecosystems are shown as the

Theme	Question
Dynamic strategy devel- opment and implementa- tion	How can strategies and implementation be cocreated with mul- tiple stakeholders, and what is the impact of these cocreation processes on the firm and its stakeholders?
Market, economy and complexity economics	How can S-D logic incorporate various schools of economic thought and serve as a foundation for developing a theory of the market and the economy as a precursor to a more general theory of society? How can concepts from complexity eco- nomics be used to develop a general model of a complex ser- vice ecosystem that could then be used to further research on markets and the economy? Will cognitive assistants/mediators, such as WATSON, as they become part of intelligent service systems, improve or hinder the decision-making of marketing personnel and consumers?
Big data	How can Big Data be used to capture actor-centric behavior and provide the means to calibrate nonlinear, dynamic models of market actors (e.g., suppliers, firms, customers) in a service ecosystem?
Macromarketing: ethics, environmental sus- tainability, social sustainability, and public policy	Can S-D logic and institutional theory be used to investigate the process and types of institutional innovation that could fos- ter ethical decision-making? How can S-D logic and ecosys- tem service(s) be used to advance environmental sustainabil- ity? How can S-D logic, informed by ecological theory, be used to understand issues of social sustainability? How might public policy be modified to benefit society by encouraging col- laboration and cooperation among firms in national and global service ecosystems, and what governance (institutional) safe- guards would be necessary?

TABLE 1.3. RESEARCH FRONTIERS OF SERVICE-DOMINANT LOGIC

most significant entity, containing social and economic dimensions. The ES is related to three dimensions, and the value creation cascade operates across all of them.

Matthies *et al.* also define the term "value-in-impact as a spatially and temporally dynamic component of value-in-use and value-in-exchange, which represents the co-creation and co-destruction of potential value (positive and negative impact) attributed by actors to how ecosystem services are managed, facilitated, and utilized by human-based service systems". In other words, the ES can be considered a service system where the actors are living beings by themselves (i. e., a forest or a lake), which provides offerings with a potential value that human beings can create or destroy. Finding a clear explanation of the relationships and behaviors between actors and operand resources in



Figure 1.4: A service-dominant value creation (SVC) framework for ecosystem service offerings in value co-creation within the socio-ecological system (figure taken from [86])

an ES is not easy. In recent years, there have been many contributions in computing systems to help understand the behavior of economic and social systems as system dynamics (SD) and agent-based modeling (ABM).

1.4. System Dynamics and Agent-Based Modeling

Nowadays, it is possible to observe different complexity levels everywhere: macrosystems like economic, social, and environmental environments; metasystems like firms, schools, or governments; and micro-level systems like human cells or chemical interactions inside our body. The interactions between the elements or actors of these systems can be direct or indirect. Each actor accomplishes one function and interacts with one or more actors directly and through others indirectly inside the same system or other systems that could be on the same or different level.

These interactions make the system behave in a dynamic form, and this behavior is not perceptible for analysts who try to explain the system's performance. The complexity of the dynamic behavior will also depend on the abstraction level and the feedback loops between system actors, which can be individuals, groups, firms, or nations. In recent years, with the help of computers, it is possible to represent the systems' complexity with certain levels of abstraction and analyze their behavior in the short or long term, for example, System Dynamics (SD) and Agent-Based Modeling (ABM).



Figure 1.5: The modeling process of System Dynamics (figure taken from [120])

1.4.1 System Dynamics

"System dynamics is a computer-aided approach to policy analysis and design. It applies to dynamic problems arising in complex social, managerial, economic, or ecological systems, i. e., any dynamic system characterized by interdependence, mutual interaction, information feedback, and circular causality" [123]. In 1961, Jay W. Forrester published the book Industrial Dynamics, where he proposed the notion of this methodology. It is now applied in economics, public policy,

environmental studies, defense, theory-building in social science, and its home field: management [46].

The modeling process of System Dynamics (SD) includes five iterative steps: (1) problem articulation, (2) dynamic hypothesis, (3) formulation, (4) testing, and (5) policy formulation and evaluation (see Fig. 1.5). Iteration occurs from any step to any other step (indicated by the interconnections in the center of the diagram). The most crucial step is problem articulation. In this step, the modelers define the reference modes and time horizon. Next, in the second step, they formulate a dynamic hypothesis, map a system structure, and chart a model boundary. In the third step, a simulation model is formulated with equations, parameters, and initial conditions. SD uses the notation of flows and levels proposed by Forrester [46]. The fourth step consists of comparing the simulated behavior of the model to the system's actual behavior. Finally, the policy design includes creating new strategies, structures, and decision rules [120].

1.4.2 Agent-Based Modeling

"Agent-based modeling (ABM) has been used to study socio-technical systems. In ABM, a socio-technical system is modeled by decomposing systems into some heterogeneous entities, called agents, which continuously interact with each other and their surrounding environment. The global behavior of these systems is the result of interaction between agents and environment" [60]. "Agent interactions are defined by a set of decision-making rules to interact with each other and the environment" [122].

Schieritz and Miling [116] compared these approaches, SD and ABM, summarizing their characteristics in Table 1.4. They concluded that "an integrated approach possibly can help decisionmakers develop the capacity of thinking of both the forest and the trees." In other words, by integrating both methodologies, it is possible to simulate complex systems on several levels. Simultaneously, these authors have used both methodologies to design and implement hybrid models. They propose different schemes for connecting the SD and ABM blocks depending on abstraction and system hierarchy.
TABLE 1.4. SYSTEM DYNAMICS VS AGENT-BASED MODELING SIMULATION [SCHIERITZ2003MODELING]

Property	System Dynamics	Agent-based Modeling
Basic building block	Feedback loop	Agent
Unit of analysis	Structure	Rules
Level of modeling	Macro	Micro
Perspective	Top-down	Bottom-up
Adaptation	Change of dominant structure	Change of structure
Handling of time	Continuous	Discrete
Mathematical formula- tion	Integral equations	Logic
Origin of dynamics	Levels	Events

1.5. Conclusions

After analyzing the concepts and principles of a circular economy, we realized that implementing it is not an easy task. It depends on many characteristics, such as the process itself, the willingness of people, institutional policies, and monetary resources, among others. Furthermore, implementing a circular economy in an established process or system requires understanding the system itself, the interactions between the elements, the complex behaviors, and the non-linearities. According to the CE principles, it is necessary to break paradigms to find new forms of maximizing a product's life to preserve the natural ecosystems. One of the principal objectives of servicedominant logic is to find new forms of value-creation to benefit the elements of a system. The next step of this research is to define a new framework that combines CE and S-D logic for establishing a new methodology for the CE implementation in a Mexican system and, later, implement a hybrid simulator based on SD and ABM. This simulator will allow us to represent the system's dynamic behavior and find new policies for adopting CE.

2. Case Study and Methodology Proposal

This chapter explains the first case study, "Jalisco sin Hambre," and how we built the methodology to design and implement a simulation model from CE strategies. First, we briefly explain the food banks' background and how they are related to the ITESO University. Next, we analyze related literature to explain how the SDL, ES, SD, and ABM interact in different applications. Then, we describe the nine-step methodology proposed. Finally, we implement the first two methodology steps in the case study.

2.1. Case Study: "Jalisco sin Hambre"

ITESO is a university entrusted to the Jesuits community. Its main objective is the construction of a more fair and human society. Its mission is deployed in three aspects:

- a) Form competitive, accessible, and engaged professionals disposed to put their being and job into social service.
- b) Expand the knowledge frontiers and culture to the endless searching for truth.
- c) Propose and develop viable and pertinent solutions for the transformation of systems and institutions [63].

ITESO has implemented many social projects with a relevant impact on the environment, society, and vulnerable people [94]. The 2030 Agenda for Sustainable Development is determined to end hunger, achieve food security as a matter of priority, and end all forms of malnutrition [27]. Food security is the state where all people have physical, social, and economic access to sufficient, safe, and nutritious food that meets their dietary needs and food preferences for healthy and active life [41]. According to the Global Food Security Index (GFSI), in 2017, Mexico had a 65.8/100 score occupying 43rd place, below Uruguay, Costa Rica, Brazil, and Argentina [30]. The poverty index of CONEVAL [23] (Consejo Nacional de Evaluación de la Política de Desarrollo Social) considers other indicators: per capita income, educational backwardness, health services access, social

CHAPTER 2. CASE STUDY AND METHODOLOGY PROPOSAL

security access, household quality, household essential services access, food access lack, and social cohesion grade. The 2016 CONEVAL poverty index indicated 24.6 million persons in poverty in Mexico. Similarly, the food access lack index was 20.1%, and 15.4% for Jalisco.

In Jalisco, to attend to these poverty and food access situations, there are five food banks located in different cities: Guadalajara, Zapotlanejo, Tepatitlán, Juanacatlán, and Atotonilco. They operate independently from each other, collaborating with local farm producers to collect food that might go to waste and deliver it to people in food insecurity. However, the producers' donations are not sufficient, and in terms of management, it is necessary to improve the food collection and delivery. In addition, it is also vital to reduce the number of people in food insecurity and ensure policies to support the food banks' viability.

On the other hand, "Jalisco sin Hambre" (JSH) is a non-profit organization created to formulate a comprehensive and replicable model that will systemize the processes of food collection, storage, conservation, and distribution to make them more efficient. The organization's purpose is to reduce the number of people with food insecurity in Jalisco. The participants in this project are from different organizations in Guadalajara, Mexico: the Jalisco state government, academic institutions like ITESO (Instituto Tecnologico y de Estudios Superiores de Occidente) and Tec de Monterrey, social organizations like ProSociedad and Germinar, technology institutions like CIATEJ (Centro de Investigación y Asistencia en Tecnología y Diseño del Estado de Jalisco) and CONACyT (Consejo Nacional de Ciencia y Tecnología), and private sector organizations like Amdocs.

JSH has three main strategies related to food insecurity: (i) operation and management improvement, (ii) environment management, and (iii) beneficiaries' development. The first strategy includes a new logistics model, the implementation of a geo-referenced system, an improvement plan for food management and safety, a strategy to encourage donations, a platform to make good use of donations, and the installation of a food processing plant. JSH is a pertinent case for applying our proposed methodology in the sense of the CE perspective because of the opportunity to use the waste, in this case, Food. Food banks are already installed; the simulator will focus on the first strategy, operation and management improvement of the food banks.



Figure 2.1: Circular economy principles proposed by Webster [138]

2.2. Obtaining the Methodology Proposal

The CE principles proposed by Webster [138] (see Fig. 2.1) are the basis for organizing the concepts. We have assigned Service-Dominant Logic to principles 1 and 5 due to the theory of service innovation and value co-creation concerning how different actors utilize waste during its life. Building resilience through diversity (principle 2) is assessed using ABM since each actor is considered an individual agent within the organization with its characteristics, and the interactions between actors are defined in an agent-based model. The third principle: "work towards using energy from renewable sources," is related to ecosystem services. In [86] explain the ecosystem as an actor providing services to societies: thus, the non-renewable sources of energy are equivalent to those ecosystem actors.

Finally, regarding the fourth principle, "think in systems," Webster suggests that organizations or firms are conceived as complex non-linear systems with diverse interactions among their elements. In this sense, system dynamics proposes analyzing systems in search of an explanation for the system's behavior (see Fig. 2.2).

According to the assignment of the five concepts, we defined six connections to analyze the relevant literature (see Table 2.1).

CHAPTER 2. CASE STUDY AND METHODOLOGY PROPOSAL



Figure 2.2: Circular economy adoption from its principles assigning the studied methodologies and tools.

Considering the literature review, we summarized that the methodology proposed is based on: (i) CE flows through a value chain (see Fig. 2.3). This configuration proposed by Kalmykova et al. [67] is used for explaining the different strategies implemented in every circular value-chain stage. These authors collected the literature concerning CE and classified the data contained in strategies proposed in the articles. (ii) SDL replaces the concept of a supply chain with a network concept that is referred to as a service ecosystem. A service ecosystem is a structure of coupled value proposing social and economic actors interacting through institutions and technology to coproduce service offerings, exchange service offerings, and co-create value instead of delivering or adding value. In the services ecosystem, there is a strong focus on collaborative processes [76]. (iii) ABM design applied to ES, as [90] implemented for a complex system related to land use. And (iv) The Sterman's SD modeling process, which is iterative, it is necessary to redefine some features related to other steps [120].

The methodology steps are shown in Figure 2.4. The first four stages are considered as previous work, and the following five steps are those proposed by Sterman's (2000) modeling process (see Table 2.2).

In the services ecosystem, there is a strong focus on collaborative processes [76]. (iii) ABM

TABLE 2.1. LITERATURE RELATED TO THE INTERCEPTIONS BETWEEN METHODOLO-GIES

Methodologies Authors		Main Contribution
S-D Logic and ES	Matthies et al. [86]	Value construction and potential value
S-D Logic and Lusch and Tay [77], Ra- japakse and Terano [106], [127], [105]		ABM in a service industry, ISPAR, NKCS, and customer engagement model
ABM and SD	Schieritz and Milling [116], Cherif and Davidsson [20], Borshchev and Filippov [11], Nava et al. [53], Teose et al. [126], Lektauers et al. [71], Ferrada and Camarinha- Matos [44], Martin and Schlüter [85], Swinerd and MacNaught [122]	Comparison between ABM and DS, ABM and DS for CAS, Examples us- ing ABM and DS, Hybrid models
ABM and ES	Sun and Müller [121], Brady et al. [14], Filatova et al. [45], Bagstad et al. [7]	ABM with BBN and ODM for land- use decisions, ABM with AgroPoliS software, ABM for SES, ABM for ES flows.
ES and SD Boumans et al. [12], Bouman et al. [13], Arbault et al. [3]		GUMBO model, MIMES framework, LCA on GUMBO model
S-D Logic and SD	Qiu [103]	Dynamic model of a service system

design applied to ES, as [90] implemented for a complex system related to land use. (iv) The Sterman's SD modeling process, which is iterative. It is necessary to redefine some features related to other steps [120]. The methodology steps are shown in Figure 2.4. The first four stages are considered as previous work, and the following five steps are those proposed by Sterman's (2000) modeling process (see Table 2.2).

Each step has its process. From step five to step nine, we propose that the modeling process is iterative in all senses. For example, in building the simulation model in software (step 7), it would be necessary to review a preliminary step if we found inconsistencies. We have chosen a case study





Figure 2.3: Circular economy resource flows through a value chain (taken from [67]).

developed in our community to validate our methodology: "Jalisco Without Hunger." This project addresses food insecurity in the Guadalajara Metropolitan Area.

2.2.1 Select interacting parts of the value chain

The value chain elements are connected to interact for a value co-creation between them. Lusch [76] claims that S-D logic replaces the concept of a supply chain with a network concept that is referred to as a service ecosystem. A service ecosystem is a structure of coupled value proposing social and economic actors interacting through institutions and technology to co-produce service offerings, exchange service offerings, and co-create value instead of delivering or adding value. In the service ecosystem, there is a strong focus on collaborative processes.

The first step of our methodology is to choose two or more actors in the CE value chain or service ecosystem that interact among themselves. For example, we will choose the seventh element



Figure 2.4: Methodology proposal for implementing a hybrid-service simulation model for CE

of the CE value chain: recycling and recovery. According to Fig. 2.3, this entity interacts with collection and disposal, remanufacturing, and materials sourcing. We select two entities: recycling and recovery (R&R) with collection and disposal (C&D). In the JSH case study, the main actors are the donors as the C&D part and the food banks as the R&R part.

2.2.2 Analyze and choose the CE strategies for each part or entity

According to Porter [101], it is required to find the intersection points between the organization and society where the CE strategies have the more significant impact. For the entities selected, R&R and C&D, Kalmikova et al. [67] made a CE strategies summary and found some implementation cases in four strategies related to food waste. R&R's strategies are by-products use, cascading materials, and restoration. For C&D, the strategy found is logistics/infrastructure building.

By-product utilization refers to products from other manufacturing processes, and their corresponding value chains are used as raw materials for manufacturing new products. A concrete case, for instance, is the reuse of grain residues as food for fish. We can find cascaded uses in the food and beverage processing industry for cascading materials in the food sector, and restoration in the CHAPTER 2. CASE STUDY AND METHODOLOGY PROPOSAL

TABLE 2.2. STEPS OF THE METHODOLOGY PROPOSAL TO IMPLEMENT A CE SIMULA-TION MODEL

Step	Description
1	Select interacting parts of the value chain
2	Analyze the CE strategies for each part or entity
3	Define the entities' attributes according to the CE strategies
4	Define the agent-based model for the selected entities
5	Define the dynamic behavior or a mathematical model of the environment variables
6	Map a diagram connecting the entities or agents with the environment
7	Build the simulation model in a specialized software
8	Define and execute performance tests
9	Formulate and evaluate policies

food waste sector, requires establishing requirements to set up composting and digesting. According to the agricultural food sector, logistics service providers should invest in tools to optimize the information link between retailers and consumers to assess consumer needs more accurately.

One of the main strategies in the JSH case study is the operation and management improvement, which includes: a new logistics model, a georeferenced system implementation, an improvement plan for food management and safety, donations encourage strategy, donations exploitation platform implementation, and food processor plant installation. This strategy can be compared with those presented by Klamikova [67].

2.2.3 Define the entities' attributes according to CE strategies

The attributes entities in the JSH case study can be defined individually for each actor. For producers, the attributes are status score, location, and food characteristics like type, quantity, and expiration date. The food bank provider (FBP) gives the status score producer, and it depends on the previously donated food conditions. This score is helpful for FBP because he will know before

accepting the donation and spending economic resources if the food donation is in good condition for assembling food boxes that later will be delivered to needed people.

For food banks, the attributes are location and economic resources availability. There is another entity in the interaction between producers and food banks: the transporter. The transporter must pick up the food at the producer's location and take it to the food bank. The transporter's attributes are travel duration, travel price, and available schedule.

2.2.4 Define the agent-based model for the selected entities

In order to define the agent-based model, it is essential to define the interactions between entities. There are several interactions between the three entities defined before in the donation process: producer, food bank, and transporter. When the producer is aware that some food remains, he registers the donation attributes in a donation platform. The food bank is notified about the possible donation, and the food bank provider (FBP) analyzes the donation attributes, mainly the producer status score. This status score is the priority for accepting or rejecting the donation. If the producer score is good, the FBP asks for the transportation availability sending a request with the producer attributes and FB location. The transporter answers with a notification containing the travel duration, travel price, and schedule. Next, the FBP answers the transporter with the transportation acceptance or rejection and, at the same time, confirms to the producer about the donation acceptance and the transporter schedule.

2.3. Conclusions

The present chapter established the basis for proposing a methodology for the hybrid simulator implementation. We also presented the case study "Jalisco sin Hambre", where the simulator will be implemented. We explained the JSH importance for Guadalajara's metropolitan zone and the magnitude of the food sector in the CE context. Following the proposed methodology, we defined the first four steps at the JSH entities: producers, food banks, transporters, and the interactions between them. In the next chapter, we will describe the visit to the food bank in Tepatitlán, Jalisco, to observe the FB functions and dynamics.

CHAPTER 2. CASE STUDY AND METHODOLOGY PROPOSAL

3. "Jalisco sin Hambre": Tepatitlán Food Bank

In chapter 2, we proposed a methodology for implementing a simulation model to evaluate Circular Economy strategies based on four perspectives Service-Dominant Logic (SDL), Ecosystem Services (ES), System Dynamics (SD), and Agent-Based Modeling (ABM). Also, we introduced the current situation in Mexico regarding food insecurity. We briefly explained the "Jalisco sin Hambre" organism and how the food bank (FB) operates in a general form. In order to collect actual and accurate data, we visited three food banks: Tepatitlán, Guadalajara, and Zapotlanejo. We observed their functioning and interviewed the FB directors to obtain a better understanding. For simplification issues, we chose the Tepatitlán FB (TFB) for implementing the methodology proposed.



Figure 3.1: The fifteen regions attended by the Tepatitlán Food Bank in Jalisco.

3.1. Food Bank Process

The operation of the food banks is as follows: Food Banks' value chain begins with producers that have some food available. They communicate with the food bank manager, indicating a possible donation. The food bank manager sends a truck to pick up the food donation, and the food bank pays the logistic' expenses. When the food donation arrives at the food bank, the food in good condition is separated. Later, it is classified and packed to be delivered to needy people, who pay a minimum. This money is utilized to cover the food bank's operating expenses.

This TFB was established in 2007. It delivers to 4300 families (19,500 people) a food bag that contains a mix of vegetables, fruits, grains, bread, and tortillas, to each family at a low cost every two weeks. The family density is about 4.7 people per family. Besides the families, the TFB attends rehabilitation centers, orphanages, public dining rooms, and the patient's family at the Regional Hospital of Tepatitlán. They deliver the food bags to people from fifteen different regions (see Fig. 3.1) in three different colors: green for vegetables and fruits, orange for semi-processed food, and red for highly-processed food. The price of the red bag is higher than the others. This configuration was chosen for teaching people which food is the healthiest.



Figure 3.2: Total food received at Tepatilán FB during 2017 (kg).

Besides the food bags, the TFB delivers other services: the mobile kitchen, vegan cooking classes, purifying water for tortilla production, a bakery, and four rehabilitation centers; dental

care; psychological care; and home gardens workshops. The TFB process begins with the food bank provider (FBP), who is in charge of contacting the producers for food surplus. If the producer has available food, the FBP will send a truck to pick up the food; the whole food received during 2017 can be seen in Fig. 3.2.



Figure 3.3: Tepatitlán food bank process, from the food donations until the food bag delivery to families.

TFB pays the transportation expenses. When the food arrives, after being unloaded, it is inspected for separating the bad conditions food from the rest in order to form the food bags. When the food bags are completed, they are uploaded to the transportation and delivered to the families previously selected for receiving the help (see Fig. 3.3).

3.2. Methodology Steps

The following paragraphs will describe the modeling process based on the proposed methodology [56] applied to our case study "Jalisco sin Hambre" (JSH).

3.2.1 Choose interacting parts of the value chain (entities)

The circular value chain for biological cycles [82] is shown in Fig. 3.4. The biological processes are different from technical processes because of the product nature. In biological processes, the product has an expiration date to be consumed. As described in the TFB process, we have chosen two entities from the circular value chain in this context are the farmer and the product manufacturer. In the JSH case study are the producer and the FB provider.



Figure 3.4: Circular value chain for biological cycles (obtained from [82]).

3.2.2 Analyze the CE Strategies for each part or entity

Kalmikova et al. [67] analyzed several circular strategies from different authors and products and services. Recently we found other authors who propose circular strategies for avoiding food waste [95]. These strategies are ordered from those who deserve more attention to the little attention (see Fig. 3.5). The best strategy for food waste reduction must be prevention and avoiding surplus food generation throughout food production and consumption. The next one is to re-use surplus food for human consumption for people affected by food poverty through redistribution networks and food banks. The third strategy is to recycle food waste into animal feed and composting. The fourth strategy is to recover energy: e.g., via anaerobic digestion. Moreover, the last strategy is to dispose of unavoidable waste in engineered landfills with a landfill gas utilization system. The TFB attends the second and third strategies.



Figure 3.5: Food waste hierarchy strategies (taken from [95]).

From this analysis, the main objective of the hybrid service simulation model (HSSM) is to generate and provide information to policymakers about the advantages of implementing circular economy strategies in the short and long term. The HSSM specific objectives and boundaries are defined from the JSH strategies and the service dominant logic (SDL). The HSSM will be capable to:

- a) Visualize the improvement in the TFB logistics and donations to the food banks
- b) Quantify the environmental impact of the food waste generated for the FB

- c) Verify the policies defined for implementing circular economy strategies in order to improve donations and economic resources
- d) Quantify the value generated by the JSH actors in the FB processes.

These boundaries will allow us to define the entities attributes according to the CE strategies for step three.

3.2.3 Define the entities attributes according to the CE strategies

From selecting of the entities described in step 1: producers and food bank provider, it is essential to understand how they interact with each other and the decisions that each one make. The food producers usually calculate the food production according to the market demand, season, and land and weather conditions. If the producer has a food surplus production, they usually give it to the cattle owners for animal consumption; in Tepatitlán, there are also dairy and meat producers.



Figure 3.6: Three layers meta-model. (taken from [43]).

The task of the food bank provider is to contact the producer for a possible donation or sale at a lower price for the food surplus production. If the producer decides to donate the food, the food

bank provider will send transportation to pick up the food. When the food arrives at the food bank, the operators separate the good-condition food from the bad-condition food; the later is separated as food waste, and a composting organization collects it.

3.2.4 Define the agent based model for the selected entities

To define the agent-based model, we first define the meta-model based on [43] and [26], who define a multi-agent system as having six essential elements (see Fig. 3.6): (a) an Environment, E; (b) a set of objects (artifacts) that exist in E; (c) a set of agents, A; (d) A set of relationship, R, which defines the relationship between objects and agents; (e) A set of operations, O, that agents can use to affect objects; and (f) a set of universal law which determine the reaction of the environment to agent operations.

Furthermore, it is also important to define the hierarchy among different environment levels and distinguish between social agents and non-social agents [60].

Meta-Model Element	Element in JSH
Environment, E	Emissions to air, food waste deposition, weather conditions, rainforests
A set of artifacts that exist in E	Land-use, cattle
A set of Agents, A	Social level: producer, FB director, social worker, beneficiary; Operational level: food bank providers
A set of relationships, R	Social level: Producer – FB Director – Social worker – Beneficiary; Operational level: Producer – FB Provider – FB Director – Social worker – Ben- eficiary
A set of operations, O	Food production, Food bank collection and separa- tion, Food bags delivery
A set of universal laws which deter- mine the reaction of the environment to agent operations	Food waste deposition will deplete the land.

TABLE 3.1. IDENTIFICATION OF THE META-MODEL ELEMENTS IN JSH

According to our case study JSH we have defined the elements of the meta-model in Table 3.1. The entities selected will be agents: producers and FB provider. Also, we identified three other

CHAPTER 3. "JALISCO SIN HAMBRE": TEPATITLÁN FOOD BANK

agents who interact with them directly and indirectly: the FB director, the social worker, and the beneficiary.



Figure 3.7: Causal diagram of the natural environment related to food production.

There are three main processes to be modeled at the operational level: food production, FB operations, and food bags delivery. The agents will decide on these three processes.

3.2.5 Define the dynamic behavior of the environment variables

In the previous step, we defined the environmental variables as emissions to air, food waste deposition, weather conditions, and rainforests availability. The only data available is the food waste generated from the TFB during 2017, which in total was 105,123.3 kilograms.

3.2.6 Map a Diagram connecting the entities or agents with the environment

We used a causal diagram to define the relationship between environmental variables and food production (see Fig. 3.7). At the highest level, we find the rain that influences the growth of the forest positively and, at the same time, the land use conditions. The more forest growth increases, the density of the forest will be higher. If the density of the forests grows, the land-use conditions will be better, then the surface of the soil to be sown will be greater. However, if this surface increases, the growth of the forest will decrease. At this point, we find the first balance cycle, B1.



Figure 3.8: Interaction among environment variables and food production, using stocks and flows.

The second balance cycle, B2, is described by the variables related to land-use surface, which will allow that food growth increases, and at the same time, the food produced will increase. Nevertheless, food production will decrease the land-use conditions. The balance cycle B3 begins with the food-produced variable. Food consumption will also increase if this increases, and food waste will grow. The food waste deposed on the soil will decrease the land-use conditions. This causal diagram has been translated into flows and stocks (see Fig. 3.8) as system dynamics methodology suggests. In this representation, we are already considering the agent Producer.

3.2.7 Build the simulation model in a software

The TFB operation has been modeled using system dynamics with stocks and flows (see Fig. 3.9). In this model, we considered two types of food: produced and semi processed food. The produced food comes from the farmers directly, and the semi-processed food is bought for the food banks to complete the food bags for the families. We are already contemplating the decision makers, even if they are not programmed as agents. Also, the food waste stock is missing.

The software used for the model construction was Anylogic [2], the advantage of using this package is that system dynamics and agent-based modeling can be programmed in the same system. In the next step, we will present the first performance tests for this model.

CHAPTER 3. "JALISCO SIN HAMBRE": TEPATITLÁN FOOD BANK



Figure 3.9: Tepatitlán food bank operations using stocks and flows.

3.2.8 Define and execute performance tests

The first performance test for the food bank operations considers the stock's initial values equal to zero, and the values of the variables are defined in Fig. 3.10. The FB food at the beginning is growing until it stabilizes, which means that the food produced is equal to the food consumed by the families. It is also possible to observe that the agents are considered parameters so that agent programming will substitute these parameters. Independently of these performance tests (see chapter 5), we have investigated some CE policies related to food waste deposition.

3.2.9 Formulate and evaluate policies

We studied the Life Cycle Assessment (LCA) approach [8] to evaluate policies. This analysis quantifies the energy used and environmental impact on the air, water, and soil during its life cycle (see Fig. 3.11). From its raw materials until its final utilization, this study requires many data about the chemical process, i.e., the chemical compounds emitted to the air, water or soil.

According to the TFB, the energy used is gas for the bakery and tortilla processes, oil for trans-



Figure 3.10: Tepatitlán food bank operations using stocks and flows.

portation, and electricity for the food bank operation. Air emissions and food waste are from transportation, freezing, tortilla production, and bakery processes. From this analysis, we realized that the particular emissions measurements for each process do not significantly impact in the food bank analysis, as we are focusing on the food waste treatment of the food bank as a whole. On the other hand, in August 2018, the TFB installed a food processor plant (FPP) to transform into puree and juices the food that will soon expire. This FPP will be added to the daily operation of the food bank. The analysis of this process is pendant as future work. In this part of the policies evaluation, we found some waste management alternatives [65, 84]. These alternatives are:

- a) Anaerobic digestion: is a biochemical pathway able to convert almost all biomass sources (including wet materials such as organic wastes and animal manure) to biogas.
- b) Composting: is a natural process where microorganisms and fungi decompose organic material into a humus-rich soil product (compost).
- c) Incineration: is the controlled combustion of the waste material with a surplus of air. It is used for treating heterogeneous waste such as mixed food and packaging waste.

CHAPTER 3. "JALISCO SIN HAMBRE": TEPATITLÁN FOOD BANK



Figure 3.11: Life Cycle Analysis elements for any process (taken from [8]).

- d) Landfilling with landfill gas collection: landfill gas generation is the result of anaerobic degradation of organic matter.
- e) Dry/wet feeding: is employed for livestock animals, and it is another alternative. Some food requires heat treatment to reduce the risk of foreign animal diseases and eliminate possible harmful pathogens.
- f) Pyrolysis and gasification: is a thermochemical process that, by heating in the absence of oxygen, converts organic material to solid, liquid, and gaseous fractions of charcoal, bio-oil, and biogas, respectively.

3.3. Conclusions

In this chapter, we first described the Tepatitlán food bank operation in more detail, given some operational numbers. Then, we defined the hybrid service simulator model (HSSM) and the boundaries or specific objectives that the simulator must meet. Additionally, we presented the first iteration of the proposed methodology. The purpose is to classify collected information regarding the JSH project, specifically the Tepatitlán food bank, and present some modeling assumptions for the simulator construction. As we mentioned in the second chapter, the construction of the simulator is an iterative process; in this work, the first iteration is described for obtaining the simulator.

In the construction of this first iteration, some pending tasks were identified. The first is to define and program the agent-based model for the actors identified in the meta-model. The second

task is to add the food waste stock to the food bank operation model. In third place, it is crucial to define the performance tests of the model. Finally, it is important to define and model the food waste policies in the simulation model.

Chapter 3. "Jalisco sin Hambre": Tepatitlán Food Bank

4. Agent-Based Modeling

4.1. Introduction

A system modeled as a collection of autonomous decision-making entities or agents is known as an agent-based modeling system. Each agent individually assesses its situation and makes decisions based on specific rules, and agents may execute various behaviors appropriate for the system they represent [10].

"The emphasis on modeling the heterogeneity of agents across a population and the emergence of self-organization are two distinguishing features of agent-based simulation compared to other simulation techniques such as discrete-event simulation and system dynamics." [79]

Agent-based modeling (ABM) captures emergent phenomena resulting from individual entities' interactions. Application of ABM in environmental issues is diverse [14, 45, 59, 109, 121]. On the other hand, there are studies of ABM implemented in products and services systems [106]. Lusch and Vargo suggest using computing tools to find the explanation of complex service ecosystems [135].

A typical agent-based model has three elements: (i) a set of agents, (ii) their attributes and behaviors: a set of agent relationships and methods of interaction, and (iii) the agents' environment, i.e., agents interaction with their environment in addition to other agents [79].

Furthermore, a computational engine for simulating agent behaviors and interactions is needed to make the model be executed successfully. In other words, to have agents repeatedly execute their behaviors and interactions.

However, simulating a supply chain system using systems dynamics modeling (SDM) takes a continuous-time approach and constructs models with differential equations. SDM involves stocks as a representation of state variables and flows as the rate of the stocks changes over time. Flows get in or out of stocks and can be represented continuously or at discrete time points [120].

This chapter defines the agent-based model for the case study *Jalisco sin hambre*. We utilized the modeling sequence described in the Modeling Agent systems based on Institutional Analysis (MAIA) meta-model proposed by [49]. Furthermore, We have replaced the Physical Structure,

which describes the Food behavior using SDM.

4.2. The MAIA Meta-Model in the case study JSH

The MAIA model describes a methodology to construct an agent-based model from any sociotechnical system (STS). It is organized into five structures or categories: Collective Structure (agents and their attributes), Constitutional Structure (the social context), Physical Structure (the physical aspects of the system), Operational Structure (the dynamics of the system), and Evaluative Structure (the concepts that are used to validate and measure the outcomes of the system) [49]. The author defines and describes an agent-based model according to these five structures. In the following sub-sections, we explain how each structure functions and the valuable concepts for our case study JSH.

4.2.1 Collective Structure

The Collective Structure (CS) describes the characteristics of the community composed of agents. The agents in the simulation all take an agent type, and they are individual or composite entities that make decisions, act and react in a social system. According to [49], an agent type is described with the subsequent attributes: name, properties, personal values, information, physical assets, possible roles, intrinsic capabilities, and decision-making criteria.

According to the case study JSH, we describe six different agent types: *producer*, *FB provider*, *FB director*, *distributor*, *social worker*, and *beneficiary* (see Table 4.1). We realized that all the agents manage the same physical component: Food.

The *producer*'s properties are Food surplus, Food type, and quantity, and his values are geographical distance and the Status given by the Food bank providers. This agent requires information from the environment and physical structure like market demand, Food production, and Food expiration date.

Its principal decision-making criterion (DMC) is Food Donation (see Table 4.2). There are some conditions to be considered prior to Food donation. First, the Food bank (FB) provider must contact the *producer* to ask him if there is a Food surplus in his Food production. If the Food surplus exists, the *FB provider* must collaborate with the Food transportation according to location,

Agent	Properties	Personal values	Information	Possible role	Physical Component	Decision- making criterion
Producer	Food sur- plus, Food type, Food quantity	Geographical distance, Status	Market de- mand, Food production, Food expira- tion date		Food	Food dona- tion
FB provider		Location	Food Bank Stock, Number of benefited families		Food	Contact pro- ducers
FB director		Money for buy- ing Food	Food Bank Stock, Number of benefited families		Food	Contact distributors Distributor
Food type	Geographical distance, Lower Food price, Status	Market de- mand, Retailer price, Food expiration date		Food	Food selling price	
Social worker			Number of benefited families	Food deliverer, the Family in- terviewer	Food	New family addition
Beneficiary	Age, Gen- der, Com- munity Id, Location, Telephone	Geographical distance, Money for buying Food bags	Delivery dates	New candi- date, Active, Inactive	Food	Delivery at- tendance

TABLE 4.1. COLLECTIVE STRUCTURE OF JSH AGENT MODEL

transportation time, and capacity. If this decision tree is actual, the *producer* donates the Food. The value of the *FB provider* is the location. He will require information about the number of benefited families and the Food available in the Food bank. The DMC is Contact Producers to obtain Food donations (see Table 4.2). Before contacting the producers, the *FB provider* must review if the Food available in the Food bank is sufficient for covering the Food requirements of the beneficiaries. If there is still missing Food, the *FB provider* will choose a *producer* to contact him and review his Status. If the *producer* Status is greater than 7, he will decide to call him. The *producer* Status is a score given to producers according to the Food delivered in previous donations. The larger the score, the smaller the Food waste.

No.	Rule name	Condition	Consequence
1	Producer Surplus	Food Surplus > 500 kg AND Contact date < Food lifetime	
2	Food trans- portation	Location is in route = True AND Transportation time < 1 day AND Transportation capacity > Food surplus	Transportation available = True
3	Producer do- nation	<i>FB provider</i> Contact = True AND Food available = True AND Transportation available = True	Food donation = True
4	Producer contact	Food required > Food Goal AND Producer Status ≥ 7	<i>FB provider</i> Contact = True
5	FB director contact	Food required > Food Goal	Contact Distributor = True
6	Food pur- chase	Available money ≥ Food Total Price AND Contact Distributor = True	Food purchase decision = True
7	Set retail price	Required Food \geq 10 tons AND Payment type = cash	Retail price = Food Total Price
8	New family addition	Food access frequency < once a day AND Family study Status < 0.5	Family active = True
9	Transport Food	Vehicle for Food = True AND Cost of vehicle < \$20	Food bag transportation = True
10	Delivery at- tendance	Food bag transportation = True AND Delivery dis- tance < 3 km AND Money for Food bag ≥ \$150.00	Go for Food = True

TABLE 4.2. DECISION-MAKING CRITERION

The third agent is the Food bank (FB) director. His value is the money for buying the Food. He will require the same information as the *FB provider*: the number of benefited families and Food available in the Food bank. His DMC is Contact Distributors, for buying the required Food at the lowest prices. First, he will check the Food needed for completing the Food requirement. Then he will contact the *distributor* and will ask for a Food quotation. Then the FBD will revise if there is available money for buying Food. If there is enough money, he will decide to buy the required Food.

The *distributor* is the fourth agent, and his properties are the Food type available, Food quantity, and the Food expiration date. His values are geographical distance, the lower Food price, and the

Status given by the Food bank. The information that the *distributor* requires is the market demand and retailer price. Moreover, this agent's decision is related to the Food selling price. There are a few pre-conditions that the *distributor* has to consider before setting the retail price: the quantity of Food required and the payment type. The first one must be greater than 10 tons and the second if the payment is in cash. If both conditions are actual, the *distributor* will set the food price and communicate with the food bank director.

The person who deals with the beneficiaries is the *social worker*, the fifth agent. We considered that this agent does not have properties and personal values that are important for the model's objective. The information required for the *social worker* is about the number of families that will be benefited. In contrast to the previous agents, the *social worker* performs two roles that are important to the system behavior: he is a Food deliverer every 15 days, and also he is the family interviewer that will decide if a new family will be added the Food security program.

For adding a new family to the program, the *social worker* must ask about the Food access frequency of the family. We have considered it less than once a day. If this condition is true, the *social worker* collects information about the family in a sheet (see Table 4.2), like the number of family members, if the family chief has a job, the family chief's age, and calculates the nutritional index of the family members. Then they are added to the list of beneficiaries who receive the Food bags every 15 days.

The last agent that we defined is the *beneficiary*, and his properties are age, gender, community, and telephone. His values are geographical distance and money for buying the Food bags. The *beneficiary* needs to know the delivery dates. His possible roles are new candidate, active *beneficiary*, and inactive *beneficiary*.

The attendance of the Food bags delivery is the decision he needs to make. The first condition is to revise if there is transportation for the Food bags. A Food bag weighs around 30 kilograms; it is crucial to have some vehicle to transport the Food bags: a bicycle, tricycle, wheelbarrow, car, or truck, depending on the distance between the *beneficiary*'s home and the delivery location. The following condition to check is the available money for buying the Food bag price: 150.00 pesos. If this condition is met, the *beneficiary* will go for the Food bag.

4.2.2 Constitutional Structure

The Constitutional Structure is a collection of roles, institutions, and dependencies. To be part of a social system, agents take roles and behave according to specific institutions. The institutions are social rules, norms, or strategies that influence behavior and decision-making [49]. The formal definition for institutions is the ADICO institutional statement and contains five components: Attributes, Deontic, aIm, Conditions, and 'Or else' (ADICO). Attributes are any value of a person to whom the institutional statement applies. Deontic can be one of the three modal verbs using deontic logic: may (permitted), must (obliged), and must not (forbidden). Aim describes the particular actions to which the deontic is assigned. Conditions are answered by the questions when, where, how, and to what extent an aim is permitted, obligatory, or forbidden. 'Or else' define the sanctions imposed for not following the rule. The shared strategies can be written as AIC (attributes, aim, conditions); all norms as ADIC (attributes, deontic, aim, conditions); and the rules as ADICO (attributes, deontic, aim, conditions, or else) [24].

Six institutions were identified for the Food bank process in the case study JSH (see Tables 4.3 and 4.4). Food reception, Food collection, Food delivery, Food bought, Food price, and Food conditions. The Food reception institution is attributed to the *beneficiary* and is an obligation to attend Food deliveries always. If the *beneficiary* misses twice is out of the program. Due to this institution having a sanction is considered a rule.

Food collection is an institution attributed to the *FB provider*. It is an obligation to pick up Food from donators from Monday to Thursday, as this institution does not have any sanction (or else component) is a norm. Similarly, the Food delivery and Food reception institutions are norms attributed to the *social worker* and *FB director*. The institution for Food delivery is an obligation to deliver Food bags to the beneficiaries twice a month, which is considered a norm. The Food bought is permission to deal the Food price with the *distributor* until the Food price is less than the retail price, and is considered a shared strategy. The Food condition is attributed to the *FB provider*: is "an obligation to separate Food in bad conditions", and is considered a norm because it does not have any sanction.

Role/Agent Objective		Institution	Entry Condition
New candi- date/ <i>beneficiary</i>	Have enough Food to feed his family		Contact social worker
Active/beneficiary	Attend the Food deliv- eries every 15 days	Food reception	Attendance to Food de- liveries
Inactive/beneficiary			Do not attend to Food deliveries two times
Family interview- er/social worker	Register <i>beneficiary</i> characteristics		Contact beneficiaries
Food Deliver- er/social worker	Deliver Food every 15 days	Food delivery	Deliver all the Food bags every 15 days
FB provider	Get Food to achieve Food goal	Food collection, Food conditions	Food required for the FB stock
Get money to buy Food and pay FB peopleFood b price		Food bought, Food price	There is no economic resources
Producer	Income		Not income expected
Distributor	Income		Not income expected

TABLE 4.3. CONSTITUTIONAL STRUCTURE

4.2.3 Physical Structure

The physical structure specifies relations between physical components in a composition diagram and a connection diagram [49]. For our case study JSH, we identified only one physical component: Food (see Table 4.5). Its properties are type, weight, expiration date, purchase price, and status. As all the agents in the model have access to it, the Food is considered a public component. Its behavior is to feed people. Moreover, according to the MAIA model, its affordances (potential uses) are: produced, sold, bought, donated, processed, semi-processed, packed, delivered, and wasted.

On the other hand, System Dynamics (SD) analyzes system behavior through the definition of flows and stocks. In a supply chain, the stocks represent containers of a quantity of some product,

Name	Role	Deontic	Aim Condition		Sanction	Institution type
Food reception	Active beneficiary	Obligation	Attend Food deliveries	Missing attendance ≤ 2	<i>beneficiary</i> is out of the program	Rule
Food collec- tion	FB provider	Obligation	Pick up Food from donors	From Monday to Thursday	-	Norm
Food delivery	Food deliverer	Obligation	Deliver Food bags	Once every 15 days	-	Norm
Food bought	FB director	Obligation	Get required Food	$\begin{array}{ll} \mbox{Food} & \mbox{ac-}\\ \mbox{quired} & \leq \\ \mbox{Foodrequired} \end{array}$	-	Norm
Food price	FB director	-	Deal the Food price	Food price < Retailer price	-	Shared Strategy
Food condi- tions	FB provider	Obligation	Separate Food	Is Food in lousy condi- tions?	-	Norm

TABLE 4.4. INSTITUTIONS

which can only be accessed by flows influenced by external variables [120]. We defined the Food affordances as stocks and the flows between stocks as actions to be executed to fill or empty the stocks (see Figure 4.1).

The stock called "Produced" refers to the Food produced by the agent *producer*, and this stock is emptied by two flows: *demand* and *donation*, which refers to the market demand and Food donation, respectively. The demand flow fills the "Sold" stock, whereas the donation flow supplies the "Donated" stock. The donated Food is transported (*transport*) to the "FB Food" stock. At the same time, the "Semi processed" stock refers to perishable Food, which is "Bought" by the *FB director*

Name	Properties	Туре	Affordances
Food	Type, Weight, Expiration date, Purchase price, Sta- tus	Public	Produced, Sold, Bought, Donated, Processed, Semi-processed, Packed, Delivered, Wasted

TABLE 4.5. PHYSICAL STRUCTURE



Figure 4.1: Physical structure represented as flows and stocks in a system dynamics model.

to complete the Food required to deliver to the beneficiaries. This Food bought is also transported to the FB. From the FB Food stock is emptied by two flows: *waste* and *packing*. The first supplies the "Wasted" stock, and the second supplies the "Packed" stock, which is the Food divided in bags that is "Delivered" (the last stock) to the beneficiaries.

4.2.4 Operational Structure

The operational structure describes the dynamics of the agents' influence on the system's state concerning time and space. It defines the order of the agents and physical components' actions. In every time step in the simulation, each agent enters the action arena to explore the actions he may execute. The action arena is defined in each simulation by a list of action situations, which at the same time, are described by several related entity actions that take place using plan specifications. There are four plan types: atomic plan, which consists of a single entity action; sequence, which is a set of actions that will be executed in the specified order; choice, which is a set of actions from which one is selected randomly (with equal probability); and loop, which is a plan that is repeated for as long as a condition holds [49].

The action arena for our case study is described in Table 4.6. We defined six action situations. The first one, Producers Contact, which the *FB provider* performs, is a loop conformed by two entity actions: review the FB stock, which means to compare the current level of the FB stock with the Food required to attend to the beneficiaries. The second action is to review the *producer* Status

if this is greater or equal to 7. This action situation will maintain until the FB stock is greater or equal to the Food required. Until the loop condition is reached, the post-conditions will be changed: increase FB stock and decrease the number of producers contacted, and the flag *FB provider* contact is set to True.

Action situation/ Performer	Plan type	Entity actions	Rule name	Loop condi- tion	Post-condition	Institution
Contact producers/ FB provider	Loop	Review the FB stock, Review <i>producer</i> Status	<i>producer</i> contact	FB stock ≥ Food re- quired	Increase FB stock, Decrease producers, <i>FB provider</i> contact = True	-
Donate Food/ producer	Sequence	Review <i>FB provider</i> contact, Review Food Surplus, Re- view Transportation	<i>producer</i> donation	-	Donation Status = True, Update Food donation character- istics: Food type, weight, expiration date	-
Buy Food/ FB director	Sequence	Review FB stock, Review retailer Status	Food pur- chase	-	Increase FB Stock	Food Bought
Transport Food/ FB provider	-	Review donation Status, Review donator location, Calculate trans- portation expenses	Food trans- portation	-	Set transportation time, Transporta- tion Status = True	-
Separate Food/ FB provider	Sequence	Check Food condi- tions	Is Food in good con- ditions?	FB stock is empty	Decrease FB stock, Increase packed Food stock, In- crease Food waste,	-
Deliver Food/ social worker	Atomic	Deliver Food	FB stock = Food re- quired	-	Increase Food de- livered, Decrease Food packed	Food delivery, Food reception

TABLE 4.6. ACTION SITUATIONS

The second action situation, Food Donation which the *producer* performs, is a sequence of three entity actions: (1) review if *FB provider* has called him; (2) quantify if the Food surplus exceeds 500 kg, and establish the Food expiration date; (3) review the transportation availability, i.e., if the transportation is in route at the current day, and the transportation capacity exceeds the Food

surplus. The post-conditions for this action situation are: the donation Status is set True, and update the Food characteristics: Food type, weight, and expiration date.



Figure 4.2: Entity actions related to the flows in the physical structure.

The third action, Food Bought, which the *FB director* performs, is a sequence that consists of two entity actions: (1) review the FB stock and (2) review the retailer Status. This sequence will be executed until the FB stock exceeds the Food required. This action situation will be regulated by the institution Food Bought. The post–condition will be to increase the FB stock. The Food Transportation, the fourth action situation performed by the *FB provider*, follows a sequence of entity actions conformed by: (1) checking donation Status, (2) reviewing donor location, and (3) calculate transportation expenses. The pre-conditions for this action situation are: the donation situation be true, i.e., that the *producer* has accepted to donate the Food; and that the *producer*'s location of the is on the Food collection route. The post-conditions are: to set the transportation time and to enable the transportation flag.

The Food Separation, the fifth action situation which the *FB provider* performs, is a loop that will be executed until the FB stock is empty. The sequence will follow two entity actions: (1) to check Food conditions, and (2) to separate Food. The Food conditions are defined with two values: good and bad conditions. The post-conditions are: to decrease FB stock, increase pacing Food, and increase Food waste. The sixth action situation is Food Delivery, which the *social worker* performs. This action is an atomic plan because it is performed just once every 15 days. The entity's action
related to this situation is to deliver Food. The pre-condition is that the Food required is the same as the Food collected in the FB stock. The post-conditions are to increase the Food delivered stock and to decrease the packed stock.

This set of action situations is related to the set of flows in the physical structure (see Fig. 4.2). The Buy Food action will enable the *buying* flow. The Transport Food action will attend the flows *transport* and *transport* flows. The Donate Food action will activate the *donation* flow. The Separate Food action will regulate the *packing* and *waste* flows. Finally, the Deliver Food action will enable the *delivering* flow. The relations of each agent with their respective actions are shown in Fig. 4.3 and the relations between them.



Figure 4.3: Agent interaction with their physical structure.

4.2.5 Evaluative Structure

The Evaluative Structure defines those indicators that show patterns of interaction, evaluation, and outcomes, where it is possible to observe discrepancies between simulation and reality due to software bugs or conceptualization errors. These indicators show the model validity (is it realistic?) and the model usability (can we prove policies?) The variables that review the states of the

system and its performance are called problem domain variables, which sometimes are related to variables in the model. The validation variables are used for debugging and validating the model [49]. According to the case study, the scope matrix of the evaluative structure is shown in Table 4.7. The variables defined are obtained from the stocks of the Physical Structure, and in this table, these variables are related to the entity's actions influence on those variables. If the influence is direct, we will use a 'd', and an 'i' if the influence is indirect.

Entity actions/ Problem domain variables	Review the FB stock	Review producer Status	Contact <i>producer</i>	Review Food surplus	Review transportation	Review retailer Status	Review donor Status	Review donor location	Calculate transportation expenses	Check Food conditions	Separate Food	Deliver Food	Review Food market demand
Kg of Food in the FB	d	i	i		i		i	i		i	d		
Kg of Food produced													d
Kg of Food sold				d									d
Kg of Food bought	d					d			d	d	d	i	
Kg of Food donated	d	d		d	d	d	d	d	d	d	i	i	d
Kg in the FB stock	d				d		i	i	i	d	d	d	
Kg of Food wasted		d					d			d	d		
Kg of Food packed										i	d	i	
Kg of Food delivered											i	d	
producers contacted	i	d	d		d		d	d	d				
Transportation time					d				d				

TABLE 4.7. SCOPE MATRIX OF EVALUATIVE STRUCTURE

4.3. Conclusions

This first approximation of the agent-based model gives us a clear idea of the agent's interaction and decisions. Furthermore, how the agents interact with their environment, in this case, Food is the physical component. The MAIA Meta Model describes five structures for defining the agent attributes and performance. The FB responsible has validated this agent model, at a conceptual level. The next step is to choose and program the model. Once the model is programmed and tested, it is important to validate it in terms of replicability in the real world. Later, we will apply the Circular Economy policies and measure the circularity of the process.

5. Building the Simulation Model in Netlogo

5.1. Introduction

This chapter mainly explains how the agent-based model is translated to the software. We have chosen NetLogo software [139] to implement the case study because it is an open-source program, and it allows to design and connect a hybrid model (ABM and SD). Besides, we present the ODD (Overview, Design concepts, and Details) protocol [104] to define some qualitative characteristics of the model. Finally, we present some graphs to show the model performance. Additionally to the model simulation, we investigated some information related to the circular economy policies and how to evaluate them.

5.2. ODD Protocol

Agent models (AM) are not easy to design, and it is essential to keep in mind all the model's characteristics. There are some approaches dedicated to describing the agents' characteristics and behaviors. The ODD (Overview, Design concepts, and Details) protocol, proposed by [104], is designed to create a model description that is easy and quick to read and understand. The protocol comprises three general elements: overview, design concepts, and details; these elements group seven characteristics that every AM developer must describe.

The overview part involves three elements: (1) purpose, (2) entities, state variables, and scales, and (3) process overview and schedule. The design concept is considered the fourth element, which describes the characteristics of the model like basic principles, emergence, adaptation, objectives, learning, prediction, sensing, interaction, stochasticity, collectives, and observation. Finally, the details are shaped by (5) initialization, (6) input data, and (7) submodels [104].

5.3. Tepatitlán Food Bank ODD Protocol

Now, we describe the ODD protocol for the Tepatitlán food bank (TFB) case study. This description will be helpful for programming the AM.

5.3.1 Purpose

The purpose of the model is to evaluate, in the short and long terms, environmental decisions in the TFB case study in order to transform it from a linear supply chain to a circular one and implement a services ecosystem between agents, finding a value generation between actors.

5.3.2 Entities, State Variables, and Scales

The model includes six agents or entities: producers, *FB providers*, *FB directors*, *distributors*, *social workers*, and *beneficiaries*. *Producers* have attributes transformed in variables for food surplus, food type, food quantity, location, status, food production quantity, food expiration date, and food donated quantity. *FB providers* have attributes for location and review the status of FB stock quantity and the number of benefited families. In practice, food has been considered a physical component, and it is represented using System Dynamics, like a supply chain (see Fig. 5.1). Each stock is considered a state variable, and the agents will manipulate the flows between each stock. We also have two other state variables: families benefited and food bank money. The time step is fifteen days, which is the frequency for delivering the food bags to the families, and the food unit is kilograms.

5.3.3 Process overview and scheduling

The food bag delivered to *beneficiaries* is composed of 50% of vegetables and fruits, which are donated directly from *producers*; 30% corresponds to grains and cereals (beans, rice, sugar, and soy); and the other 20% are sanitary products and groceries donated by The National Association of Food Banks (BAMX).

The process begins with buying the percentage of food like grains and cereals; this action is performed once every fifteen days. The second action, performed by the *FB providers*, is to contact the *producers* to get the food donations. The food donation occurs if three conditions are satisfied: (i) the existence of food surplus from the *producers*' inventories; (ii) the food lifetime is enough to handle it and deliver to *beneficiaries* (greater than one week); (iii) the transportation is available. In other words, the transportation capacity is greater than the food donation, and the Producer location is on the route from other *producers*' donations.



Figure 5.1: System dynamics model for Tepatitlán food bank.

When the food donated arrives at the food bank, the food in good conditions is separated from lousy conditions, considered waste. The remaining good-condition food is packed in individual bags with the grains and cereals, sanitary products, and groceries for delivery to the *beneficiaries* (see Fig. 5.1).

5.3.4 Design concepts

There are eleven different design concepts that not all the models employ. Some models are so simple that they only describe a few [104].

a) *Basic Principles*. The first concept "Basic principles" describes the general concepts, theories, and hypotheses involving the model's design. The Tepatitlán food bank describes a typical linear supply chain that involves a physical element of food for families in food insecurity. This process involves food waste that can be used as a resource for other processes to obtain some value, like biogas or composting, for benefiting families and society. The number of families will "pull" the entire process, as the food bags obtained depends directly on this quantity. In other words, if the number of families increases, the amount of needed food will increase, and it will be necessary to obtain more resources for attending to these families. These theories are better observed in a causal diagram (see Fig. 5.2)



Figure 5.2: Causal diagram for Tepatitlán food bank.

- b) Adaptation. The agents will have an adaptive behavior since they will review some conditions in the system to make decisions. For example, the *producers* need to review if they have surplus food for donation to the food bank. The food bank Producer will seek donations from *producers* to cover food goal determined by the families benefited.
- c) *Objectives*. The main objective of the simulation is to get enough food for attending the families with food insecurity. However, as the food banks are non-profit organisms, they also need to get economic resources to operate the food bank and complete the food quantity. Some of these resources are obtained from selling the food bags to the families at a very low cost. Others are obtained from direct donations of money. The circular economy policies pretend to use food waste to obtain some money or other value to obtain more food and economic resources. To measure the objective achievement, we monitor some variables: food bank stock, money, families benefited, and food waste.

- d) *Sensing*. The agents, especially the *FB providers*, are in charge of sensing some variables: (i) the amount of food collected, which is represented in the system dynamics model by the food bank stock (see Fig. 5.1), (ii) the *producers*' location that is available for food donations, (iii) the number of families to be attended, (iv) the food conditions donated by *producers*, and (v) the food lifetime. The *social workers* sense the families' attendance to pick up the food bags. The families evaluate their resources to gather up the food.
- e) *Interaction*. The model agents interact with each other directly and indirectly (see Fig. 5.3). The *FB provider* interacts directly with *producers* and *social workers*. The *FB director* interacts with the food (grains and cereals) *distributor*. The *social worker* interacts directly with *beneficiaries* and *FB providers*. At the same time, all the agents interact directly with food as the physical component acting as flows regulators. The *FB provider* decides over the fruits and vegetables, waste, and packing flows. The *FB director* affects the grains-and cereals flow. The *social worker* determines the delivery, new families, and deserters flow. The *producers* decide the amount of production.
- f) Stochasticity. To represent the causes of variability, we model the producer's production with random numbers. Similarly, the addition of new families and deserters is simulated with random quantities.
- g) Collectives. Collectives represent producers and families. Producers shape a collective determined by the number-producers slider. This number will never change during the simulation. The families also are represented by a collective initially defined by Num-families slider; however, this number will be modified over the simulation execution by the new families and deserters flows.
- h) Observation. The most critical variables to observe are food bank stock (FB-stock), food required, and food waste. The first two variables are plotted on the same graph to see if the food goal has been met. In addition, as the simulation setup is executed, we can observe the *producers*' and families' configurations.



Figure 5.3: Agent interaction with their physical structure.

5.3.5 Initialization

The model initialization is divided into the system dynamics model (SDM) and the agent model (AM). The initialized variables in an SDM are the levels or stocks (see Table 5.1). The AM requires defining some particular attributes of each agent and the associated variables (see Table 5.2).

5.3.6 Input data

The input data has not been defined yet. The objective is to use information related to food production, food seasonality, population growth, household food waste, water and energy employed in the Tepatitlán community, and the food bank.

Variable	Value
FB-Money	MXN \$1,000,000.00
food bought	food required = food bag weight * Num families * (1 + waste index)
Production	Initial production = sum[food produc- tion] of <i>producers</i>
BAMX	1,000 tons
FB-Stock	0 tons
food-wasted	0 tons
Packed	3000 tons
Delivered	0 tons
Familes-benef	270 families

TABLE 5.1. INITIALIZATION OF SYSTEM DYNAMICS MODEL

5.3.7 Submodels

NetLogo allows simulating SDM and AM. The AM is a programming language that defines all the agents' attributes and submodels. It is essential to define the main function, called "go", where all the processes will be executed. Based on the MAIA meta-model [49], in the chapter 4 we defined the Action arena, which is the core of the "go" function. In the following paragraphs, we describe the processes and how they were codified.

SDM execution. The SD model is considered a submodel, and it is executed every step time because the conditions are constantly changing; in [139] is possible to find the instructions to execute it.

Calculate the food required. The food required (foodr) depends directly on the number of families (numfam), the food bag weight (fbagw), and the waste index (wasteidx). These three variables can be modified on the user interface; then we can say that the food required is determined by the

Agent	Quantity	Shape	Color	Size	Position	Own variables
Producers	Number of <i>pro-</i> ducers	Plant	Green	1.5	Random	food production = Nor- mal(5000, 1000); food sales = Normal(4000, 500); food lifetime = Poisson(3); Loca- tion route = false; Status = Poisson(8)
FBproviders	1	Person	White	2	Center (0,0)	-
FBdirectors	1	Person	Blue	2	-	Money collected = 100,000
Families	100	House	White	1	Random	Miss attendance = 0; Family status = Normal(0.7, 0.1); food access freq = Poisson(3); food bag transport = True; Delivery dist = Normal(1, 0.1); Money for food = Poisson(300)

TABLE 5.2. INITIALIZATION OF AGENT-BASED MODEL

user indirectly:

$$foodr = numfam * fbagw * (1 + wasteidx)$$
(5.1)

Buy. In this sub-process, the grains and cereals are bought. The quantity of bought food (foodb) depends directly on the food required (eq. 5.1), and corresponds to the 30% of this quantity defined by the food bank. As this action is executed once a month, and the time step is 15 days, the formula calculates the food required for two deliveries in one month (eq. 5.2). This food quantity is stored in the Bought stock in the SDM (see Fig. 5.1).

$$foodb = 2 * 0.3 * foodr \tag{5.2}$$

Donate food. When this process is executed, it initially sets false the food available and transportation available variables. The second step is to establish the location of the *producers* running the "locate" sub-process, which will be explained below. The third step is to review the three con-

ditions required for the food donation from *producers* (see chapter 4). The first condition is related to the producer status, determined by previous donations. If this status is greater than 7, the producer contact is true. The second condition will evaluate the producer food for donation. There are two attributes to evaluate the food surplus and the food lifetime. The food surplus is calculated by subtracting from the amount of food production minus food sales (eq. 5.3).

$$food \ surplus = food \ production - food \ sales$$
 (5.3)

The food lifetime is a characteristic established by the producer. If the food lifetime is longer than seven days and the food surplus exceeds 500 kg, the available food variable is true. The third condition to evaluate is food transportation, and the FB has its trucks for transporting the food. The only condition to evaluate is that if the producer location is on route for food collecting with other *producers*, the available transportation variable is true. Then when the three variables: producer contact, food available, and transportation available, are true, the donation amount is equal to the food surplus of the producer. Then the amount of food production for each producer is recalculated (eq. 5.4).

$$food \ production = food \ production - food \ sales - food \ surplus$$
 (5.4)

When all the *producers*' donations are available according to the conditions mentioned above, it is important to sum up all the individual *producers*' donations. The "need completing variable" is false if this sum is greater than the food required for vegetables and fruits, corresponding to 50% of the food required. This variable will be evaluated later in the Grain and Cereal sub-model. Before closing the Donate food process, we randomly program the food production from *producers* for the next step, and the Producing flow from SDM will be updated.

Locate. This sub-process will evaluate a location route among the close enough *producers* for picking up the food. NetLogo allows asking for the nearest agents in a certain radius. If these nearest neighbors (near-neigh) are greater or equal to 2, then the variable location-route is true.

Grain and Cereal. In this process, the "need completing" boolean variable is evaluated. If if this is true, it means that the fruit and vegetables donated by *producers* are not enough, and the food that is missing will be completed by the grains and cereals. To calculate the percentage of

the missing food (completing-index), we divide the number of fruits and vegetable donation (f-vdonation) between the food required (foodr), and it is subtracted from the 50% that the fruits and vegetables would cover (eq. 5.5). Then the flow of grains and cereals is recalculated as shown in equation 5.6. Otherwise, the grains and cereals are calculated with the 30% previously defined.

$$Completing index = 0.5 - (fv \ donation)/foodr$$
(5.5)

$$Grains and cereals = (Completing index + 0.3) * foodr$$
(5.6)

Groceries donation. This amount corresponds to 20% of the food bag composition (eq. 5.7). The national food bank association BAMX directly donates the groceries. The type of groceries generally are cleaning products, food cans, and soda.

$$Groc \ donation = 0.2 * foodr \tag{5.7}$$

Family submodel. The families are modeled every step with specific attributes: (i) the number of times that the family misses a food bag delivery, (ii) the status of the family according to the socio-economic study, (iii) food access frequency, (iv) if they have transportation for carrying the food bag, the distance from their homes to the delivery place, (v) and if they have enough money for paying the food bag. In this submodel, there are two sub-processes: "add family" and "remove family". The first process creates ten families each time step, review if their status is less than 0.5 and the food access frequency is less than 2, then the family is added to the families benefited stock in the SDM through the "new families" flow; otherwise, they are eliminated. The "remove family" process only reviews if the family has missed two food bags deliveries consecutively. Then the deserters flow from the SDM model is modified with the number of families that comply with this condition and are also eliminated.

Deliver. When this process is executed, the food bag has been delivered and paid to the *social workers*. This money is collected in the FB-Money stock, and later this money will be used for paying the food bank operating costs.

Do plot. This process automatically updates the graphs according to the actual values of the variables each time step.

This formulation in ODD format allows us to describe the agent-based model of the Tepatitlán food bank. It is helpful, and the authors employ NetLogo in their publications for showing examples using this format [104].

5.4. First Results of Model Execution

As we have described before, we are calculating some variables' values as random distributions for preliminary tests. The user interface is divided into four parts: controlling buttons, variables sliders, the world sight, and the output variables graphs (see Fig. 5.4). There are two controlling buttons: setup and go. The setup button corresponds to the initialization element of the ODD protocol. And the go button executes all the sub-models described in the previous section.



Figure 5.4: Tepatitlán food bank simulation setup.

First, the model must be initialized using the setup button to execute the simulation. The agent model creates the agents with their attributes, and the SDM is prepared with the initial values for stocks. The user can modify the variables sliders or leave them with the previously programmed values.

When the go button is pressed, the simulation begins until the go button is pressed again to stop the simulation. The graphs are updated each step time, and as we can observe (see Fig. 5.5), the food requirement (blue) grows as the number of families grows (orange), the food collected by the food bank (black) is bigger than the food requirement. Finally, the food waste (green) is accumulated in a continuously growing stock. According to the causal diagram (see Fig. 5.2), the

positive or reinforcement loop generated from the families benefited is replicated in the simulation model.



Figure 5.5: Tepatitlán food bank simulation execution.

5.5. Conclusions

This chapter describes the hybrid model designed in the chapter 4 using the ODD protocol and checks if the model is complete. We programmed the MAIA meta-model described by the ODD protocol in NetLogo, and we found that it is possible to combine SD and ABM in a hybrid model, which allows us to understand and separate the agents (people) from the artifacts like food (physical elements). In the preliminary results, we showed the first assumption: the number of families will make the food bank stock also grows (reinforcement loop) by analyzing the cause and effect diagram (see Fig. 5.2) and programming the hybrid model (SDM and AM) in the NetLogo software. However, the data required from the food bank is still missing, and it is necessary to get it for the model validation, which corresponds to step eight of our methodology. After model validation and analyzing the information delivered by the model, we will pass to step 9: "Formulate and evaluate policies." The constraint for this last step is that the policies must be obtained from circular proposals, satisfying the circular economy principles. Circularity indicators must be defined depending

on the validation model proofs and the case study objectives to measure this objective. The circular policies proposed will be implemented in the simulation model as future scenarios, and their economic, environmental, and social impacts will be quantified in the short and long term. As we continue applying the methodology, we have found its viability for creating a simulation model for implementing circular economy principles in Mexican industries.

Chapter 5. Building the Simulation Model in Netlogo

6. Model Validation

6.1. Introduction

In previous chapters, we explained the pertinence of circular economy in the Mexican linear processes and the importance of developing a tool like a simulator to prove some policies before implanting them in the real world. Furthermore, we visualized the tools and concepts to develop the simulator like service-dominant logic, ecosystem services, agent-based modeling (ABM), and system dynamics (SD). Then, we proposed a new methodology to design and implement a simulation model from a linear process and analyzed the case study: Jalisco without hunger. Next, we described the first attempt at methodology implementation and the concepts for representing the chosen case study: the Tepatitlán food bank. Finally, we designed the agent model applying the MAIA Meta Model and defined the agents involved in the model. Besides, we proposed the food as a physical component represented by stocks and flows like a system dynamics representation.

This chapter describes the next step for the methodology that corresponds to the eighth step: "Determine and execute performance tests." We understand that this step also refers to the model validation. In the following two sections, we will explain some tests to be performed in the model and how to validate them.

6.2. Model Evaluation

Several tests for evaluating the model include boundary adequacy, structure assessment, dimensional consistency, parameter assessment, extreme conditions, integration error, behavior reproduction, behavior anomaly, family member, surprise behavior, sensitivity analysis, and system improvement [120].

6.2.1 Boundary adequacy test

Boundary Adequacy tests assess the appropriateness of the model boundary. We constructed a causal diagram to explain the model boundary (see Fig. 6.1). This diagram can observe the

exogenous variables: food baskets per family, number of families, food bag weight, food waste index, number of producers, transportation available, and status desired. On the other hand, the model's endogenous variables are food required, food bought, food donation, food packed, and food waste.



Figure 6.1: Causal diagram for defining the model boundary of the Tepatitlan Food Bank.

6.2.2 Structure assessment test

Structure assessment tests ask whether the model is consistent with knowledge of the actual system relevant to the purpose. The model structure has been developed using the MAIA Meta-Model for agent-based modeling (ABM) [49]. This model allowed us to evaluate and define the different agents' decisions and behaviors, validated by the food bank contact. According to conservation laws, food is the only physical component; the agents manipulate and transform it during the entire process (see Fig. 5.3).

6.2.3 Dimensional consistency

This test inspects the variables' units, mainly if their dimension corresponds to the model meaning and if the equations are consistent with the model logic. For our simulator model, the units' analysis is described in Table 6.1.

6.2.4 Parameter assessment

The parameter assessment seeks the real-life meaning of each estimation. The relationship modeled by the equations must be justified by statistical or judgmental methods [120].

Monthly, registers collected from the food bank obtained some data, and other information were average estimations given by employees or food bank director, and we plotted their historical behavior (see Fig. 6.2). We analyzed the data obtained, which is summarized in Table 6.2. This table describes each variable from the causal diagram: input or output, dimensional units, sample frequency, the period that the data is available, data type, and the distribution they should follow according to the variables' nature.

6.3. Model Validation

Validation determines whether the simulation model is an acceptable representation of the actual system, given the purpose of the simulation model. There are different statistical tools for validating a model depending on the available data quantity. Kleijen considers three cases: (i) no real data available, (ii) only output data, and (iii) input and output availability [69].

In the first case, if data is not available, strong validations are impossible. At least a Sensitivity Analysis (SA) should be performed. SA is an investigation of the reaction of the simulation responses to extreme values of the model's structure. Besides, SA shows which factors are essential and how to change them in the real system if they are controllable [69].

In practice, the SA is done through a Design of Experiments (DOE). The DOE's central problem is how to select a limited set of combinations of factor levels to be observed. Next, the resulting data of the simulation experiment are analyzed by applying ANOVA or regression analysis (RA). This analysis estimates the importance of the individual factors, and the result of the RA is a model called a meta model. Typically, this meta model uses a polynomial approximation. The main objective of the SA is to find out whether the simulation model contradicts qualitative or expert knowledge. If the simulation's behavior violates this knowledge, the model should be seriously evaluated for programming and conceptual errors [69].

In the second case, the ideal simulation model would have a statistical distribution function for its output that is identical to the distribution for the real system. In practice, the manager is interested in the particular characteristics of the distribution, like the system's mean and variances [69].

In the third assumption, with accurate input and output data after running the simulation program, analysts compare the simulated output with the real historical output, and a regression model is obtained that should validate the simulation model [69].

6.4. Conclusions

This chapter describes the eighth step of the proposed methodology, which consists in "Determine and execute performance tests." We developed four tests from twelve that Sterman suggests. These tests have helped us debug the simulation model related to the mathematical equations and general performance, observing the state variables and the output.

On the other hand, validating the simulation model is aligned to specific standards found in the concerning literature. The lack of enough data from the TFB has made us search for other alternatives to validate the simulation model. We have found that sensitivity analysis can be an option to validate the simulation model by developing a design of experiments to find the most important factors or inputs of the simulation model. Once these results are obtained, we will continue with the ninth step, "Formulate and evaluate policies," which proposes circular economy policies to implement in the simulation model and evaluate future scenarios for the case study

Variables/Equations	Units
Number of Families	Families
Food bag weight	Kilograms
Food waste index	Dimensionless
Number of producers	Producers
Food baskets per family	baskets/family
Food required	Kilograms
Food bought	Kilograms
Food donation	Kilograms
Food packed	Kilograms
Food waste	Kilograms
Food production	Kilograms
Food sold	Kilograms
Food surplus	Kilograms
Food lifetime	Days
Grain and cereal food	Kilograms
Food access frequency	Times/day
Delivery distance	Kilometers
Money for food	MXN Pesos
Food bag price	MXN Pesos/basket
Missattendance	Times/month
Completing food index	Dimensionless

TABLE 6.1. VARIABLES DIMENSION



Figure 6.2: Data collected from Tepatitlán Food Bank.

Variable	Input / Output	Units	Sample Freq	Begin	End	Data type	Distribution
No. Families	Input	families	Monthly	Jan-14	Dec-18	Discrete	Poisson
Food bag weight	Input	kilograms	-	-	-	Continuous	Normal
Food waste rate	Input	9/0	Monthly	Jan-17	Apr-19	Discrete	Binomial
No. Producers	Input	producers	-	-	-	Discrete	Poisson
Production qty.	Input	kilograms	-	-	-	Continuous	Normal
Food type	Input	-	-	-	-	Categorical	-
Food baskets	Input	baskets	Monthly	Jan-14	Dec-17	Discrete	Poisson
Food bought	Input	kilograms	Monthly	Jan-17	Dec-17	Continuous	Normal
Food donation	Input	kilograms	Monthly	Jan-17	Dec-17	Continuous	Normal
Fruits and veg- etables	Input	kilograms	Monthly	Jan-14	Apr-19	Continuous	Normal
Food lifetime	Input	days	-	-	-	Continuous	Log Normal
Grains and ce- reals	Input	kilograms	Monthly	Jan-17	Dec-17	Continuous	Normal
Food delivered	Output	baskets	Monthly	Jan-17	Dec-17	Discrete	Poisson

TABLE 6.2. PARAMETER ESTIMATION

Chapter 6. Model Validation

7. Second Case Study: Confectionery Factory

7.1. Introduction

The methodology proposed for implementing the hybrid service simulation model has been designed to be applied in any company organized through a value chain. Ellen MacArthur Foundation [33] proposes a classification of this value chain: biological cycle and technical cycle. Both cycles are composed of a traditional supply chain and extra loops that close the loops and convert the linear supply chain to a circular one. The methodology proposed in this thesis consists of nine steps based on four perspectives: service-dominant logic, ecosystems services, agent-based modeling, and system dynamics. As a case study, we implemented this approach in a food bank located in Jalisco, México. Seven of the nine steps were implemented successfully, and for reasons out of our concern, the two final steps were not implemented, which are related to model validation and proposals for implementing circular economy strategies. Consequently, and to validate the methodology proposed ultimately, we applied this methodology to a second case study: a confectionery factory in Jalisco, México. This chapter will present an overview of the food industry, the new case study, and the first step of implementing our methodology, 'Select interacting parts of the value chain (entities)' [56].

7.2. Food industry

The 2030 Agenda for Sustainable Development resolved to end hunger, achieve food security as a matter of priority, and end all forms of malnutrition [92]. Food security is defined as the state in which all people, at all times, have physical, social, and economic access to sufficient, safe and nutritious food than meets their dietary needs and food preferences for a healthy and active life [129].

The current food industry seems insufficient for the global population; there are alterations in overall nutritional needs and rising economic incomes [128]. Besides, there are remarkable challenges to transforming a food system into a healthy, nutritious, and environmentally sustainable

[88], for example, water scarcity, energy use, and land availability [130], [42].

Ellen MacArthur Foundation (EMAF) published a report on the food systems and their performance in big cities. In this document, the authors explain why the current feeding systems are no longer sustainable. Some of these reasons are: the generation of large amounts of food waste generated by large cities, the way food is currently produced is harmful to human health, and the distance between cities and rural areas where food is grown is too far. Addressing this problem, EMAF proposes three ambitions that cities can realize to catalyze a circular economy for food: (1) source food grown regeneratively and locally where appropriate, (2) make the most of food, and (3) design and market healthier food products [32].

The confectionery sector is not the exception, and these challenges are essential in connecting nutrition, health, and environmental issues [88]. For example, México is second place in confectionery consumption in Latin America and sixth place globally. The estimated consumption per capita is around 4.5 kg per year [31]. On the nutritional side, in recent years, undernourishment has decreased, in contrast with obesity increase, due to factors like food consumption rich in fat, salt, and sugar. México and USA have the highest index of adult obesity, and México occupies fourth place globally in childhood obesity [72]. Moreover, the increasing consumption of confectionery products is pressuring the global supply chains. For example, cocoa and palm oil are grown in only certain places [88].

As the diversity of products, the confectionery supply chains are complex: the raw material, specialized equipment, different formulation for creating nutritious and tasty recipes, and several strategies to sell products. A system-view approach is required across the confectionery supply chain (from raw material to disposal: cradle to grave) to attend to the transition towards environmental sustainability [88]. For this reason, it is crucial to evaluate the circularity performance of the organization to assess the correct part of the system.

7.3. Circular economy analysis

The impact of CE implementation in an industry is difficult to quantify and visualize. According to [114], the circularity performance monitoring is difficult to assess as CE has a dynamic nature. In the literature, several instruments evaluate and quantify engagement and product performance

concerning CE principles [17, 39, 80, 102, 113, 137]. These instruments involve the several stages of a supply chain like design, materials, manufacturing, packing, delivery, and final user. We chose the Circular Economy Indicator Prototype (CEIP) [17] to evaluate the circularity performance of our case study. This instrument contains fifteen questions to evaluate five cycle stages of a circular supply chain: design, manufacturing, commercialization, in use, and end of use. Each question associates a measurable variable with a CE principle using a weighted score.

In the Design stage, CEIP evaluates principles like material selection, cascades thinking, and material identification. The material selection principle assesses the use of reused or recycled material that reduces waste. Also, this principle refers to dematerialization; in other words, circular design demands less material for the same performance and quality. Material identification refers to a complete bill of materials and substances in product manufacturing. CEIP uses the principles of energy identification and waste management to evaluate the circularity in the manufacturing stage. The energy identification principle evaluates the presence of a bill of energy, which provides the information required to plan for energy consumption and efficiency manufacturing processes and contributes to using more renewable energy sources. The waste management principle establishes that waste must be avoided and treated as raw material for other processes.

CEIP evaluates the cascades thinking principle in product packing and product life extension during commercialization. Besides acquisition, the diversity principle assesses the rental schemes to access higher quality products and materials without purchasing the product themselves.

'In use' circular stage assesses the cascade thinking and waste reduction principles. In the cascade thinking principle, it is essential to identify the usage status of the product that contributes to planning maintenance actions before a malfunction happens, extending the lifetime of the product. Besides, this principle identifies the importance of reusing product options enabling waste reduction. The 'end of use' stage appraises the thinking of cascades materials and waste through the availability of take-back schemes and product materials reintroduction [17].

7.4. Case Study: Confectionery Factory

We selected the confectionery factory for three important reasons: (1) it is a Mexican small and medium enterprise (SME) with around 200 employees [16], (2) its business model is a value

chain configuration [100], and (3) there is available data to be used in the simulator model. This organization has ethical values like reliability, respect, honesty, and responsibility. Unfortunately, its mission and vision do not include the sustainability, environment, or green organizations concepts. However, a confectionery recycling process is used to reuse the waste obtained from the manufacturing process, called neutral carbon filtration.

In the first chapter, we considered that our hybrid simulator methodology is oriented to companies whose environmental theme is less important than the economy because the environmental impact cannot be easily visualized [56]. In this sense, some authors question how it is possible to convince stakeholders to adopt circular economy (CE) principles in an organization [70], [138].

According to [33], confectionery manufacturing is classified as a biological cycle because the primary raw material is sugar, and the final product is edible. As CE considers the entire supply chain from the raw materials production and final disposal, we compared the supply chain of our company to that explained by [88], who established six main stages to be considered: raw materials, pre-processing (only for chocolate-based products), manufacturing, distribution and retail, consumption, and disposal. The latter two stages belong to the final user, and the other four to the confectionery manufacturer. Regarding raw materials, there are two categories: ingredients and packaging, which are inputs to the pre-processing and manufacturing processes. From this comparison, we apply the CEIP instrument described before to our case study (see Table 7.1).

We adjusted the instrument according to the product's nature. The confectionery factory only covered three of the five life cycle stages that CEIP evaluates: designing, manufacture, and commercialization. Within these elements, there are two questions that we did not consider for the evaluation: (i) "Q2: Is the product lighter than its previous version?" and (ii) "Q8: Is there a rental option for the product?". So, we eliminated their contribution in the final score, and it was calculated as shown in Table 7.1. The product rating was 52%, which corresponds to a "good" product ranking according to the CEIP scale. In the following paragraphs, we will explain how we implemented the first step of the methodology: 'Select the interacting parts of the value chain (entities)' [54].

Life cycle	Question	Answer	Scored	Rating	
Design/Redesign	Q1: Is the product made from recycled or reused material? (20p)	10%	1		
	Q2: Is the product lighter than its pre- vious version? (2p)	-	-	6/25 = 24%	
	Q3: Is there a complete bill of mate- rials and substances for the product? (5p)	Yes	5		
Manufacture	Q4: Is there a complete bill of energy for the manufacturing process? (10p)	Yes	10	25/25 = 100%	
	Q5: Is there a complete bill of solid waste for the manufacturing process? (15p)	Yes	15		
Commercialisation	Q6: What packaging is being used? (5p)	Recyclable	1		
	Q7: What is the product's warranty? (10p)	2-3 years	1	3/15 = 20%	
	Q8: Is there a rental option for the product? (15p)	-	-		

TABLE 7.1. CEIP INSTRUMENT APPLIED TO THE CONFECTIONERY FACTORY

7.5. Step 1: Select interacting parts of the value chain

Analyzing the stages that the CEIP instrument proposes, the innovation process is not frequent in the Design/redesign section, and there is not much information to be analyzed. In Manufacturing, the confectionery factory manages the confectionery waste obtained along the supply chain. We selected this stage because of the confectionery waste's recycling process, mainly composed of sugar-based products like pops and candies with their plastic packing. This recycling process is already installed in the company, and there is available data to use in the simulator model. According to [67], who analyzed the resource flows through a value chain in a circular economy, we chose the stages involved in the confectionery waste recycling process. These parts of the value chain are manufacturing, remanufacturing, recycling and recovery, materials sourcing, and design (see Fig. 7.1).



Figure 7.1: Value chain parts involved in the confectionery waste recycling process (adapted from [67])

7.6. Conclusions

This chapter presented the literature review concerning the confectionery industry in the world and México. Furthermore, we analyzed some results related to sustainability in confectionery organizations. Regarding the case study addressing, we evaluate some CE maturity frameworks. We adopted the CEIP model for measuring the current CE state of the case study. When we applied the CEIP questionnaire, we realized that questions related to commercialization, in use, and end-of-use stages, were not suitable for the food industry. Thus we found an area of opportunity for developing a CE maturity model appropriate for this sector.

According to the results obtained from this instrument and the data availability of the case study, we selected the manufacturing stage and defined the related processes according to the CE value chain. This procedure belongs to the first step of our proposed methodology, and the next steps depend directly or indirectly on this step. In the second step, we will analyze the CE strategies to

find a connection between them and select those critical for the simulator's objective and the case study.

CHAPTER 7. SECOND CASE STUDY: CONFECTIONERY FACTORY

8. Methodology Implementation in the Confectionery Factory

8.1. Introduction

This chapter explains the next steps for implementing the proposed methodology. We explain the eight steps remaining for obtaining the simulation model. In chapter 7, we performed the first step, where we selected the stages involved in the confectionery recycling process. In the following paragraphs, we will explain the subsequent steps.

8.2. Step 2. Analyze the CE strategy for each part or entity

We analyzed all the strategies [67] proposed for each stage in this step. Next, we selected those related to the food industry that the confectionery fabric had implemented.

Mainly, we observed the high-quality recycling strategy, which refers to the recovery of materials in pure form without contamination and belongs to the material sourcing stage. In the case study, glucose is recycled from the confectionery waste generated in the manufacturing process.

The second strategy observed is material productivity which refers to the economic value generated by a unit of material input or material consumption. This strategy belongs to the manufacturing process, and it is related to high-quality recycling because the recycled glucose obtained from this process is used in the manufacturing process for completing the glucose ingredient of the confectionery product.

The resulting stages selected were materials sourcing and manufacturing, which correspond to stages 1 and 3, respectively, according to [67].

From the strategies analyzed in the second step, the indicators used in the simulation model are those which explain the strategy. For the first strategy, high-quality recycling, we propose as indicators: confectionery waste, recycled material, and cost of recycled material, all in kilograms per day. For the second strategy, material productivity, we propose the same indicator, material

CHAPTER 8. METHODOLOGY IMPLEMENTATION IN THE CONFECTIONERY FACTORY

productivity (MP) measured in cost per kilogram, calculated as [93]:

$$MP = \frac{Production \ cost}{Material \ used}$$
(8.1)

The raw material used in this context is measured as the sum of virgin material (water, glucose, and sugar) and recycled material (syrup).

8.3. Step 3. Define the entities' attributes according to the CE strategies

In this step, we first defined the inputs and outputs of the stages selected in the previous step. Next, we observed the variables related to the strategies selected. Then, we defined the units of measurement in conjunction with the people involved in the industry. Finally, we obtained a list of the variables from each process: manufacturing and confectionery recycling (see Figure 8.1). These variables will be used in step seven when the simulation model is programmed [55].



Figure 8.1: Processes connected with I/O variables.

8.4. Step 4. Define the agent-based model for selected entities

The fourth step is to define the agent-based model related to the entities or processes selected using the Modeling Agent systems based on the Institutional Analysis (MAIA) meta-model [49]. In this context, we found the MAIA model helps define the agent model and their actions in different structures, like physical and institutional. In the case study, we analyzed three agents: *quality inspector, product engineer*, and *operations director*. Each one makes different decisions concerning the selected processes, and these decisions affect the processes' performance and, consequently, the variables defined in the previous step.

It is essential to define the physical component agents make decisions [49]. In this case study, the physical component refers to the confectionery product.

8.5. Step 5. Define the dynamic behavior or a mathematical model of the environment variables

This step defines the dynamic behavior using a causal diagram (see Figure 8.2). We observe two reinforcement (R1 and R2) and two balance (B1 and B2) loops. The R1 loop involves manufacturing and recycling, which both increase their production if one of them rises. The R2 loop refers to the use of coal. If the production of recycled glucose grows, it will be necessary to use more coal. In the B1 loop, the use of glucose is affected by recycled glucose, and if the recycled glucose rises, less average glucose is used as raw material. Finally, the B2 loop associates the cost of the recycling process, which will cause a decline in the production of recycled glucose.

8.6. Step 6. Map a diagram connecting the entities or agents with the environment

It is crucial to locate the agents' actions in the physical structure; this step will facilitate programming the model in the software [49, 55]. We defined three agents in step 4: the *quality inspector*, *product engineer*, and *operations director*. The *quality inspector* will decide the waste quantity according to the quality yield of the confectionery product. The *product engineer* will determine




Figure 8.2: Dynamic behavior of the confectionery case study.

the quantity of recycled glucose used in the confectionery manufacturing. The *operations director* will decide when to recycle glucose taking into consideration the cost of raw materials like sugar, glucose, and coal.

8.7. Step 7. Build the simulation model in a software

To program the model, we used version 7.1 of Vensim for Windows. We built the model from the previous steps. We modeled a system using flow and level variables as System Dynamics theory suggests [46, 120]. We identified four stocks in the model: manufacturing, confectionery waste, recycling production, and recycled glucose. The input flows in the manufacturing stock are raw materials like glucose, sugar, packing, colorants, flavorings, and recycled glucose. The manufac-

turing's output flows are the confectionery product and the confectionery waste (i.e., the product that does not meet the quality standards).

In our case study factory, there is a stock where the confectionery waste is stored until the recycling process requires it. Once obtained, the recycled glucose is stored until the manufacturing process requests some glucose to produce confectionery. In the recycling process, some waste is represented as an output flow (see Figure 8.3).



Figure 8.3: Simulation model of the confectionery case study.

8.8. Step 8. Define and execute performance tests

We executed the performance tests defined by [120] to calibrate the model. The manufacturing process is pulled by demand, around 400 kg/hr. As the confectionery waste is a portion of the production (around 4%), its behavior is proportional. The recycling process is not always on because its execution depends on three variables to be observed: (1) if there is enough confectionery waste

CHAPTER 8. METHODOLOGY IMPLEMENTATION IN THE CONFECTIONERY FACTORY

to load the recycling machine, (2) if the recycled glucose cost is less than the average glucose, and (3) if there is not recycled glucose in stock. These conditions oscillate the system behavior, and the recycling production works at only 20% of its capacity (see Figures 8.4 and 8.5).



Figure 8.4: The behavior of the confectionery manufacturing and recycling production



Figure 8.5: The behavior of confectionery waste vs recycled glucose stocks.

8.9. Step 9. Formulate and evaluate policies

According to the four perspectives proposed for assessing CE, we employ service-dominant logic (SDL) and ecosystem services (ES) to formulate policies that comply with the CE policies.

We proposed some policies described in the following paragraphs to evaluate the strategies analyzed in step 1 and the results obtained in step 8.

a) Policy 1: Marketing the recycled glucose

Since the confectionery production uses a low amount of recycled glucose, this is accumulated in stock, and the utilization of recycling production is around 20%. If half of this recycled glucose were sold to other industries, the stock would decrease sometimes, and the utilization percentage will increase to 35% (see Figure 8.6). According to CE principles, the impact of this policy must be analyzed in several approaches (economic, social, and environmental). Furthermore, the implementation of this policy requires an analysis of service logistics and marketing to take advantage of the recycling process.



MANUFACTURING VS RECYCLING PRODUCTION

Figure 8.6: The behavior of recycling production when this is sold to other industries.

b) Policy 2: Obtain confectionery waste from other industries

We propose a second policy where confectionery waste from other industries is recycled to increase recycling production. If we obtain 40 kilograms/hr of waste, the utilization percentage increases to 60% (see Figure 8.7).

These policies' economic, social, and environmental impacts must be evaluated, contrasting the organization's current situation. This analysis is out of the scope of this research. However, the simulation of these impacts is possible using system dynamics and agent-based modeling.





Figure 8.7: The behavior of recycling production when waste from other industries is recycled.

8.10. Conclusions

This chapter briefly presented the steps from two to nine of the methodology proposed and analyzed the food industry in the Mexico context. In chapter 7, we evaluated the circularity of our case study using the CEIP tool as the first step of our methodology. The results obtained were used to define the stages to be analyzed in the value chain. Then as a second step, we analyzed the strategies related to each stage and defined the entities' attributes involved.

We continued implementing each step to obtain the simulation model. We found that the recycling production oscillates due to the decisions made regarding the existence of confectionery waste, raw material costs, and operational resources. For the confectionery management, it was essential to know the utilization percentage of the recycling production process, then we considered it as our output variable. The value of the utilization changed according to the policies proposed.

9. Circular Economy Measurements

9.1. Introduction

Circular economy (CE) mainly promotes virgin material minimization and clean technologies. While CE also encourages the elimination of waste from the value chain as a benefit of reducing material cost and resource dependence, there are no specific guidelines to sectors on how to implement CE. Besides, CE omits the feature of semi-recyclability when choosing a raw material for the production process [115].

Attending this situation, in [56], we explained the pertinence of circular economy in México's linear processes and the importance of developing a simulation tool to prove some policies before implanting them in the real world: a hybrid service simulation model (HSSM). We visualized the tools and concepts to design the simulator like service-dominant logic, ecosystem services, agent-based modeling, and system dynamics. Furthermore, we proposed a methodology that consists of nine steps to follow for developing the HSSM based on circular economy strategies. These approaches allowed us to model a three-dimensional system (social, physical, and ecological) and visualize some future consequences. We prove the methodology in a first case study, where we implemented the first steps of the methodology: a food bank located in Jalisco, Mexico [55].

From the reviews obtained in the former publication [55], we implemented the methodology in a second case study, a confectionery factory in Jalisco, México. This confectionery factory recycles the caramel waste obtained from manufacturing. They use a process where the caramel is filtered, producing a syrup ready to use in the main process.

Modeling the HSSM in the confectionery factory has involved customer participation. They are operations manager, operations supervisor, recycling operator, quality manager, and costs manager in this case. There was an active contribution of all participants, which guarantees the real-world abstraction of the model. CE was a new concept for them, including the circular indicators like circularity index (CI) [25], material circularity indicator MCI [64], and circular economy indicator prototype (CEIP) [17]. For this reason, we proposed calculating the break-even point of their recycling process as an economic measure of the recycling process viability, considering that stake-

holders were familiar with this term.

Besides, we chose circularity index (CI) using the web tool called C-indicators advisor developed by [112]. It uses circular characteristics like implementation level, CE loops, CE perspective, and CE performance to propose those circular indicators suited for the particular problem attended [112]. This document describes CI and break-even point calculus, and we present the results obtained.

9.2. CE measures

In a pure sense, CE is a future in which waste no longer exists, where material loops are closed, and products are recycled indefinitely. In the actual world, that is impossible; every loop needs the energy to overcome the dissipative losses (in quantity and quality) generated in every stage of the circular stage [25].

The Ellen MacArthur Foundation describes the CE as a "new economic model which seeks to ultimately decouple global economic development from finite resource consumption" [82]. From this vision, there are two implicit premises: (1) transforming from a linear economic system to a circular one will significantly reduce the environmental impacts associated with resource use, and (2) these reductions will facilitate economic growth without environmental impacts of resource depletion [25].

From these two premises, we propose the circularity index (CI) and Breakeven-point (BP) as two simple measures to assess a circular economy implementation in a linear supply chain. We will explain these indicators in the following paragraphs.

9.2.1 Circularity index

CI is a scale ranging from linear at one end to circular at the other. This scale considers the loss of material quantity and material quality.

a) Material quantity

Conserving the material quantity means that everything that goes into the economy must equal what comes out, and there is no material loss during transformation in a perfect CE. In

practice, materials are leaked from recycling loops and require the addition of new material to maintain circularity [25].

There are two specific concerns related to material losses: material stock dynamics and dissipative material losses. The first refers to product lifetimes creating a lag between production and disposal and it takes several decades to recover material outputs that match input demands. The second explains the quantity of material recovered from product waste [25, 57]. A simple ratio (α) describes the effects of these two issues. An α value of 1 describes a perfect circularity of material quantity [25].

$$\alpha = \frac{\text{recovered end-of-life material}}{\text{total material demand}}$$
(9.1)

b) Material quality

Many recycling processes degrade and destroy the material structure, requiring energy to restore material quality. A simple ratio, β , can quantify the energy needed for material recovery relative to the energy required for primary material production from virgin resources [25].

$$\beta = 1 - \frac{\text{energy required to recover material}}{\text{energy required for primary production}}$$
(9.2)

A value of 1 for β would describe perfect circularity of material quality, an economy in which no loss of material quality occurs with each recovery cycle.

These ratios are then multiplied to obtain the circularity index (CI). Which CI = 1 refers to a theoretical circularity.

$$CI = \alpha\beta \tag{9.3}$$

9.2.2 Break-even point

The break-even point is an essential reference to the business's long-term planning, and it is helpful to know it for sales, production, operations, and investment recovery. In this sense, the break-even is when the total entry of the organization is equal to the total expenses. This mean that at the break-even point, there is no profit, or this is zero [87].

To calculate the break-even point, we use the terms: unit sale price (P) is the final value of the products or services to be sold. Variable costs (VC) are the costs that tend to vary according to the business activity level. Total fixed costs (TFC) are the costs in a period, and they are not affected by the fluctuations in the business activity levels. Turnover (Q) is the number of products produced or sold by an organization, while Contribution margin (CM) is the unit sale price less variable costs. The Break-even point (Q_e) is the product sales volume when the utility is zero [87].

$$Q_e = \frac{TFC}{P - VC} = \frac{TFC}{CM}$$
(9.4)

According to the confectionery case study, we explain the calculus of these indicators in the next paragraphs.

9.3. Case study: confectionery factory

We consider the recycling process a separate business unit of the confectionery factory; even the own managers consider this unit part of the business. To obtain the variables described by the indicators, first, we calculate the costs per kilogram of caramel recycled. Then we calculate the break-even point and the circularity index. We contemplate all the prices and costs in Mexican pesos (MXN).

9.3.1 Circularity index

We count four indicators to determine the circularity index: recovered end-of-life material, total material demand, the energy required to recover the material, and energy required for primary production. The first two indicators are required for α calculus and the other two for β . In our case study, as the caramel is a biological product, the first indicator, recovered EOL material, is the waste generated by the confectionery production, which is the 4%. On the other hand, the total material demand for caramel is around 890 kg per hour, then $\alpha = 0.04$, which measures the circularity of material quantity.

The circularity of material quality refers to the energy employed in the production and recycling

process. The confectionery process absorbs the energy required to recycle material. The energy consumed by the recycling process is around 1% of the total energy in the factory; thus, $\beta = 0.99$. In this sense, CI = 0.0396, is a low value; if the factory could get more caramel scrap from other industries, the CI would increase, as shown in Fig. 9.1.



Figure 9.1: Circularity index in the caramel recycling process.

9.3.2 Break-even point

To calculate the recycling process's break-even point (Q_e) , we analyzed their inputs like water, caramel scrap, activated carbon, sodium bicarbonate, filter helper, and paper filters. We consider the caramel scrap price the same as it is sold for other industries, like cattle breeders or beekeepers, around \$4.00. These materials, excluding caramel scrap, are around \$1.73 per kilogram. We

considered the fixed costs as \$20,000.00. The sale price (P) is \$14.70, which is the cost of the raw material for confectionery production. Then Q_e value under these conditions is 2,229.65 kg (see Table 9.1). In Fig. 9.2, we can observe four different sales scenarios in kg and break-even points. The caramel production process uses around 4,500 kg of recycled syrup per week; then, the BP is overtaken. It is important to study the total fixed costs to calculate the recycling process BP more accurately.

Variable	Value	
Sale price (P)	\$14.70	
Variable costs (VC)	\$5.73	
Total Fixed costs (TFC)	\$20,000.00	
Break-even point (Q_e)	2,229.65 kg	
Sales Q_e	\$32,775.92	

TABLE 9.1. CALCULUS OF BREAK-EVEN POINT

9.4. Conclusions

There are many circularity indicators; however, not all of them can be used in every implementation. According to the case study, choosing those that meet specific characteristics is relevant. In the confectionery case study, the reasons for adopting the circularity index and break-even point were the kind of available data and the customer's requirements to measure the Q_e of the recycling process. We calculated these indicators in the confectionery factory's present conditions and extrapolations.



Figure 9.2: Break-even point in recycling caramel process.

CHAPTER 9. CIRCULAR ECONOMY MEASUREMENTS

10. Scenario Analysis

10.1. Introduction

Due to confectionery products being diverse, the entire supply chain of confectionery products is complex: raw material supply, specialized equipment acquisition, tasty and nutritious recipe design, product packing, commercialization strategies, and the final disposal of confectionery products. A system-view approach across the confectionery supply chain is required from raw material use to the final disposal, considering a cradle to grave assessment to attend to the transition towards environmental sustainability [88]. We found significant growth in the confectionery waste valorization in the European Union. For example, the production of bioethanol and biogas as bioenergy products and bio-based and biodegradable polymers [58].



Figure 10.1: Confectionery simulator implemented in NetLogo.



(a) Manufacturing (M) vs. Recycling production (RP) processes

(b) Confectionery waste (CW) vs. recycled glucose (RG) stocks



Figure 10.2: Performance of primary variables in the hybrid simulation model.

10.2. Step 8: Performance Test Execution

For the scenario analysis, we implemented the simulator in Netlogo [139] (see Fig. 10.1). Next, we executed the performance tests defined by [120] and [104] to calibrate the model. The manufacturing process is pulled by demand, around 400 kg/hr. As the confectionery waste is a portion of the production (around 4%), its behavior is proportional. The recycling process is not operating all the time because its execution depends on three variables: (1) if there is enough confectionery waste to load the recycling machine, (2) if the recycled glucose cost is less than average glucose, and (3) if there is no recycled glucose in stock. These conditions oscillate the system behavior, and

recycling production can operate only to 25% capacity (Figure 10.2). Graph 10.2a shows that when recycled glucose is produced, manufacturing production increases. Next, graph 10.2b shows the behavior of confectionery waste and recycled glucose stocks. Finally, graph 10.2c describes the profit calculated from the raw material price and recycling total costs.



Figure 10.3: Four scenarios are proposed from two policies.

10.3. Step 9: Policy Formulation and Evaluation

According to the four perspectives proposed for assessing CE in chapter 2, we formulated policies that comply with CE principles based on service-dominant logic (SDL) [131] and ecosystem services (ES) [86] theories. Evaluating the strategies analyzed in step one and the results obtained in step eight, we proposed two policies: (i) marketing the recycled glucose; and (ii) obtaining confectionery waste from other industries to increase the recycling production. We evaluated both policies in four scenarios according to the utilization rate of the recycling process. The first scenario, entitled "low input, low output," supposed that there was a low amount of waste from other factories and a low sale of recycled glucose. The second scenario, "low input, high output," presumed that the waste acquired from other industries was low, but the recycled glucose sales were high. The third scenario, entitled "high input, low output," estimated that the waste received was low, but the sales of recycled glucose were high. Finally, the fourth scenario, called "high input, high output," estimated that the acquisition of waste and recycled glucose sales were high (Figure 10.3).

	Scenarios			
	I. LI, LO	II. LI, HO	III. HI, LO	IV. HI, HO
Confectionery waste	83.16 kg	91.2 kg	4,773.88 kg	4,773.88 kg
Recycled glucose	800 kg	800 kg	1760 kg	1760 kg
Recycling utilization	42%	42%	100%	100%
Cost recycled glucose	\$6,448.29	\$8,081.34	\$26,079.27	\$19,489.77
Benefits	\$2,432.22	\$449.34	-\$9,024.9	-\$372.3

TABLE 10.1. VARIABLES OBSERVED IN THE PROPOSED SCENARIOS.



Figure 10.4: Profit performance from the four scenarios related to the two policies: marketing recycled glucose and obtaining confectionery waste.

Several variables were proposed to evaluate the feasibility of the recycling operation as a separate business unit from the primary process. These variables are confectionery waste, recycling glucose production, cost of recycled glucose, break-even point (Q_e), recycling process utilization, and benefits. We compared these variables in the scenarios proposed (Table 10.1).

Thus, the first scenario had the highest benefits due to the cost of recycling glucose, and the recycling process utilization was 42% capacity. On the other hand, the third scenario had the low-

est benefits and the highest cost of recycled glucose due to the high confectionery waste quantity received from other industries and low sales, causing the recycling utilization to be at maximum capacity (100%) (Figure 10.4). The user determines the recycled glucose price. In the simulation results shown in Figure 10.4, we considered the recycled glucose price equal to the cost of raw material so that the use of recycled caramel could be viable. If the sale price of recycled glucose increases, the benefits of the four scenarios also increase.

10.4. Model Validation

Finally, we performed a series of interviews with recycling process decision-makers (operation manager, recycled-process owner, and quality manager) during the methodology implementation. At the end of the procedure, a questionnaire was presented, and their feedback was recorded. Key elements from the questionnaire were:

- Model purpose: the model allowed participants to understand the process behavior and their interactions. Furthermore, they identified the variables and factors that influence the entire process and how they are related.
- Model results: the model lets shareholders know the recycling process costs more precisely because the calculus was based on activity costing and not only on average costs.
- Circular economy view: the operational manager attempted to minimize the usage of the recycling process before the simulator implementation. However, after evaluating the results obtained, like costs and utilization, he decided to review the recycling process operation.
- User interface: the user interface was redesigned to be more friendly and understandable for the process owners to facilitate access to the input and output variables.
- Simulator usage: the shareholders considered outsourcing the model design service with a more straightforward user interface for daily planning during the operational process, measuring the economy's circular impact and other business variables.

10.5. Conclusions

As SDL and ES suggest, the methodology was designed to find a co-creation value between the main actors (i.e., stakeholders and environment) involved in the ecosystem. The case study implementation demonstrated through the four-scenario analysis that an economic benefit for the stakeholders could exist if the recycled glucose were commercialized. The caramel waste disposed of in landfills could be reduced regarding the environment. The process owner decided to evaluate the results shown in the simulator and consider the recycled syrup commercialization as a future project.

In the simulator, we implemented the CI and Q_e calculus. These indicators depend only on primary production, and for these reasons, they stayed constant during the scenario simulation: $(Q_e) = 6.19$, and CI = 0.14.

We found that in addition to evaluating circular economy strategies, the simulator allowed partners to understand the operation of the recycling process and visualize all the variables involved in the system. The reason is that the simulation model was developed using ABM and SD simulation methods, and understanding the system's behavior and performance is one of their main objectives.

11. Literature Review Update

11.1. Introduction

Systems modeling has been widely used to analyze and understand the behavior of complex systems. There are several methodologies and tools to obtain a dynamic model or a partial representation of a system to evaluate the consequences of some decisions before implementing them in the real world. The model is designed and computationally implemented depending on the system's scope and limitations. We can find as many systems in the world as we can observe around us. A system is "a collection of possibly interacting, related components that exhibits emergence." [119] The scientific community has developed several methodologies and theories to study systems from different points of view, depending on their application.

In this research, we applied the system theory to study the implementation of the Circular Economy (CE) concept in the real world, considering other four concepts derived from system theory: service-dominant logic (SDL), ecosystem services (ES), agent-based modeling (ABM), and system dynamics (SD). The conjunction of these tools allowed us to design a methodology for developing a simulation model for any small and medium enterprise (SME) that wanted to analyze CE strategies inside their organizations [56].

At the beginning of the Ph.D. program (2017), we did a literature review concerning these tools and concepts. After four years, we decided to do a second review to complement the former and visualize the future work regarding the methodology proposed. The following paragraphs describe the most recent research related to the tools. Furthermore, we compare the methodology proposed with a similar framework in section 11.7. Finally, we conclude with the most relevant findings and the future work associated with the research.

11.2. Circular Economy

Ellen MacArthur Foundation (EMF), launched in 2010, generates original research related to a circular economy (CE) that can contribute to solving climate change and biodiversity problems

CHAPTER 11. LITERATURE REVIEW UPDATE

by focusing on the circular economy principles implementation into practice [82]. Four years ago, when we analyzed the literature review concerning the circular economy, we found that EMF had published different reports related to this topic. In the last years, they have reported research related to artificial intelligence, electronics consumption, the future of plastics and textiles, food chain redesign, packaging, and policies, among others. [81].

The circular economy has been well adopted in the European Union (EU) by governments and private industry through the report named 'A new Circular Economy Action Plan For a cleaner and more competitive Europe' [22] published on November 3rd, 2020. However, in Latin America, there is an absence of regulations and institutions that allow circular economy implementation in the long term [9].

In Mexico, there are circumstances to analyze and deal with to implement a circular economy in the productive processes of the country. Dieleman and Martinez established some elements to explore: market and trend conditions; competitiveness and productivity; the political and regulatory framework; education, training, knowledge transfer; and learning culture [28]. Similarly, Cantú et al. identified nine categories of barriers and enablers in a literature review for a CE's successful implementation in Mexico: (1) user behavior, (2) regulatory, (3) infrastructure, (4) economy and competitive markets, (5) supply chain, (6) knowledge, (7) financial, (8) organizational, and (9) product and material characteristics [15].

Particularly, last August, Jalisco's government announced a new model for solid waste management based on CE principles, where one of their initiatives is the implementation of Integral Centers of Circular Economy (CIEC in spanish) around the state. These centers will receive urban solid waste in order to separate, treat and value it for reuse [37].

On the other hand, in 2019, an agency was created to attend to the sustainability problems in Mexico: Maken Sustanability. The organization headquarters are in Guadalajara. Its mission is "to help stakeholders make the best decisions, through international methodologies and road maps that help the company integrate social, environmental and good governance aspects to ensure sustainability [83]."

11.3. Service-dominant logic (S-D Logic)

S-D Logic has emerged as a framework that describes service as the fundamental basis of economic and social exchange [135]. Vargo et al. published a paper (2020) that summarizes the five axioms of S-D logic and discusses the conceptualization of S-D logic and other perspectives like service logic, customer-dominant logic, service science, and goods-dominant logic (see Fig. 11.1).



Focal actor(s) in value creation

Figure 11.1: Comparing the conceptualization of value and focal actors of S-D logic and related perspectives (taken from [132]).

Besides, in this paper, they identified the applications of S-D logic they call "vectors of diffusion in diverse disciplines" (see Fig. 11.2), where we can find the ecosystem services [86]. They

CHAPTER 11. LITERATURE REVIEW UPDATE

conclude this discussion with the implications of applying SD-logic: transcendence, accommodation, and transformation. They emphasize these implications in the ecosystem services point of view, sustainability as "the well-being and the viability or survivability of the system", and the governance regulations that S-D logic facilitates [132].



Figure 11.2: The broad applicability of S-D logic and existing diffusion vectors in diverse disciplines. (taken from [132]).

Regarding circular economy (CE), this year (2022), Vargo published an opinion paper comparing the goods-dominant logic, that CE proposes for the reuse of products versus service-dominant logic, where the value is obtained through the application of resources in a service-for-service exchange called a service ecosystem. These ecosystems need to be regulated by institutions to grant value co-creation [131].

Some authors have published innovation frameworks to implement the circular economy in business models using the S-D logic ([107, 118, 124]). Moreover, [50] proposes a qualitative evaluation method of service-dominant business models.

11.4. Ecosystem Services

As a discipline derived from S-D logic, Ecosystem Services (ES) has conceived nature as an actor in a service ecosystem (a forest or a lake), where it co-creates value for other actors (humans). As nature delivers services, according to the S-D logic, it needs to obtain services from the others [86].

The Consumer-Driven Business Ecosystem Research and Development (CD-BERD) model for new product and service innovations in the forest sector adopts the classical "technology-push and demand-pull" innovation model. It considers consumer values, enabling resources and dominant logic, and information flow during each phase of the research and development process leading toward new consumer-driven solutions [61].

Regarding regulations and institutions, [1] identifies and visualizes the nexuses between urban challenges (UC), ecosystem services (ES), and nature-based solutions (NBS) in order to support the mainstreaming of NBS in urban policies and sustainable and resilient urban planning interventions and strategies.

On the other hand, [18] made a literature review of ES frameworks and found that very little of the ES literature includes valuation of biophysical change (2.4%), despite many biophysical studies of services (24%). They propose as future research the integrative biophysical-social research that characterizes ES change and is coupled with multi-metric and qualitative valuation and context-appropriate decision-making.

11.5. Agent-based modeling

Agent-based modeling (ABM) has been used to explain the system behavior by observing the individual performance of the agents involved in the system. We found some research related to natural ecosystems that use ABM; for example, [97] presents a model that couples individual epiphytes' dispersal, growth, and mortality. They use substrate dynamics, obtained from a three-dimensional functional–structural forest model, allowing the study of forest–epiphyte interactions. Similarly, in [66] have designed an agent-based model to study the cultural evolution of sustainable behaviors. Besides, [91] presents a Bottom-up building stock model (BSM) based on an agent-based model eling approach (ABM) that models stock development in terms of new construction, retrofit, and replacement by modeling individual decisions on the building level.

Furthermore, [141] adopts an enterprise input-output model providing a cost–benefit analysis of industrial symbiosis (IS) integrated into an agent-based model to simulate how companies share the total economic benefits stemming from IS.

One of the most widely used tools for describing Individual ABM is the Overview, Design concepts, and Detalis (ODD) protocol [52]. Recently, the authors published a paper in which they discuss its limitations: the limited availability of guidance on how to use ODD; the length of ODD documents; limitations of ODD for highly complex models; lack of sufficient details of many ODDs to enable reimplementation without access to the model code; and the lack of provision for sections in the document structure covering model design rationale, the model's underlying narrative, and how the model's fitness for a purpose is evaluated [51].

11.6. System Dynamics

System Dynamics (SD) emerged from Philosophy, Mathematics, Logic, Biology, and Social Sciences theories. In [117], the author describes of the system theory evolution from its origins until the last findings, passing through three important threads: Dynamic and Evolutionary Systems thread, Cybernetics thread, and Soft Systems (interpretative) thread. The application fields of system theory vary from operations research, engineering and computing, economics, sociology, organization and management, political science, pedagogy, anthropology, ethics, aesthetics, semiotics, ecology, biology, medicine, psychology and psychiatry, and cognitive science (see Fig. 11.3).

The use of system dynamics (SD) in urban sustainability can be observed in [4], in which a summary of the simulation models based on SD is presented. Sustainability theories like energy, waste management, and transport are applied to design a simulation model for urban sustainability (SUSTAIN), integrating subsystems like industries, services, transport, population, and land use.

Diverse applications have been found using SD. For example, [40] presents the research ideas and conceptualization of a framework to determine the factors that influence the effectiveness of a personal healthcare response monitoring system from a systems engineering perspective. The Systems Dyanmics Society (SDS) has collected several case studies, which can be found in [123].



Figure 11.3: System Dynamics in the evolution of systems approach (taken from [117]).

11.7. Hybrid Modeling

'Modeling is essential to characterize and explore complex societal and environmental issues systematically and collaboratively. Socio-environmental systems (SES) modeling integrates knowledge and perspectives into conceptual and computational tools that explicitly recognize how human decisions affect the environment [35]." In this sense, eight challenges need to be overcome to accelerate the development and adaptation of SES modeling: bridging epistemologies across disciplines; multi-dimensional uncertainty assessment and management; scales and scaling issues; combining qualitative and quantitative methods and data; furthering the adoption and impacts of SES modeling on policy; capturing structural changes; representing human dimensions in SES; and leveraging new data types and sources [35].

Regarding ecological modeling, [78] proposes a framework for the business ecosystem model-

CHAPTER 11. LITERATURE REVIEW UPDATE

ing, including three parts and nine stages (shown in Tables 11.1 and 11.2). This framework combines business ecosystem modeling, system engineering, and ecology theories, including agentbased modeling and system dynamics.

TABLE 11.1. THE BUSINESS ECOSYSTEM MODELING WITH THE INTEGRATION OF SYSTEM MODELING AND ECOSYSTEM THEORY

Business ecosystem mod- eling	Business ecosystem	System engineering	Ecology	
Part I–Stage 1 Identify the boundary of a selected ecosystem	Domain oriented business ecosystem. Innovation ecosystem. Digital ecosys- tem.	Ontology development. System architecture	Types of ecosystem. Ecosystem hierarchy	
Part I–Stage 2 Identify actors and their roles in the ecosystem.	Stakeholders. Business models.	Domain ontology. Sys- tem standards.	Categories of organisms [74] (producers, con- sumers, decomposers.) Types of keystone species [89] (predators, mutualists, engineers.)	
Part I–Stage 3 Identify ac- tors' value propositions and business models.	Business model [142], value creation [21]	Business services. Value stream.	Ecosystem function and biodiversity [29]	
Part I–Stage 4 Identify inter- action between actors (dif- ferent types of interactions.)	Value co-creation. Value flows [86] (monetary, prod- uct, information and intan- gible.) Social network anal- ysis [5]	Flows (e.g., information exchange.) Associations (in UML diagram.) Service-oriented archi- tecture (in TOGAF.)	Intra-specific and inter- specific ecological inter- action. Ecosystem ser- vices Matter, energy and information flows.	

Comparing this framework with our methodology named Hybrid Service Simulation Model (HSSM) [56], we can find some similarities in Table 11.3. The main difference is that the HSSM does not include the ecosystem reconfiguration, which belongs to Part III - Stages 2 and 3, implementing the new proposals for the business model.

11.8. Conclusions

Four years ago, Circular Economy (CE) concept emerged recently, and there was no related research in international journals in Mexico. After this second literature review, we find some proposals presented in Mexico. This concept is essential to develop and innovate products and services through value co-creation and exchange, as Vargo suggests in a new opinion paper [131] related to

TABLE 11.2.	THE BUSINESS	ECOSYSTEM	MODELING	WITH	THE INT	EGRATION	OF
SYSTEM MO	DELING AND EC	OSYSTEM TH	EORY (CONT	.)			

Business ecosystem mod- eling	Business ecosystem	System engineering	Ecology
Part II–Stage 1 Investigate influential factors and their impact on the elements in the ecosystem (actors, roles, and interaction.)	-	Motivation and strategy (in ArchiMate.) System thinking [110] Machine logic [99]	Assessment and indica- tor of ecosystem condi- tions.
Part II–Stage 2 Investigate potential changes in the ecosystem.	Emergence and coevolution [96]	Risk management/assess- ment [111].	Ecosystem change [34]
Part III–Stage 1 Multi-agent based ecosystem simulation to identify ecosystem reac- tion towards the potential changes.	Multi-agent-based models [75]	System dynamics [68]	System dynamics, Multi- agent based modeling.
Part III–Stage 2 Ecosys- tem reconfiguration (includ- ing reconfiguration of ac- tors, roles, and interaction) due to changes.	Business ecosystem lifecy- cle [108]	System lifecycle manage- ment [111]	Evolution of ecosystems [6]

service-dominant logic (SDL). The value exchange can be given among actors like government, users, and organizations and the natural ecosystems like a forest, a lake, or the land. It is important to visualize the ecosystem services and their value delivered to the human. Agent-based modeling (ABM) has been widely used to explain and visualize the elements' interactions of a system, defining their actions and analyzing their effects on the whole system. On the other hand, system dynamics (SD) is used to observe the causal explanation in a system by analyzing its state variables.

In ecological modeling, the conjunction of these concepts, especially ABM and SD, has been widely used in recent years, as we did in our research. The analysis level obtained from these tools allowed us to design our nine-step methodology to create a simulation model which involves the scenario visualization according to the CE strategies analyzed at the beginning of the methodology. The main characteristic of this simulation model is that the final users and stakeholders can make decisions related to CE in their organization. Applying these decisions in the real world depends on several variables that are out of the scope of our methodology. However, SDL is suitable for

designing the CE strategies implementation.

For future work, first, we propose a statistical studio of the organizations that have adopted CE in Jalisco. Secondly, The validation of the methodology in other case studies to generalize its implementation. Third, design a general simulation model that would be easy to evaluate for SMEs. Finally, the design of the service ecosystem to exchange knowledge of the best practices and services that contribute to the community growth and waste minimization.

TABLE 11.3. SIMILARITIES BETWEEN HYBRID SERVICE SIMULATION MODEL (HSSM) AND BUSINESS ECOSYSTEM MODELING FRAMEWORK

HSSM	Business ecosystem modeling
Step 1: Select interacting parts of the value chain	Part I–Stage 1 Identify the boundary of a se- lected ecosystem
Step 2: Analyze the CE strategies for each part or entity	Part I–Stage 3 Identify actors' value proposi- tions and business models.
Step 3: Define and identify entities' attributes according CE strategies	Part I–Stage 2 Identify actors and their roles in the ecosystem.
Step 4: Define the agent-based model for the selected entities	Part III–Stage 1 Multi-agent based ecosystem simulation to identify ecosystem reaction to-wards the potential changes.
Step 5: Define the dynamic behavior or a mathematical model of the environmental variables	Part III–Stage 1
Step 6: Make a diagram connecting the enti- ties or agents with the environment	Part I–Stage 4 Identify interaction between ac- tors
Step 7: Build the simulation model in soft- ware	Part III–Stage 1
Step 8: Define and execute performance tests	Part III–Stage 1
Step 9: Formulate and evaluate policies	Part II–Stage 1 Investigate influential factors and their impact on the elements in the ecosys- tem. Part II–Stage 2 Investigate potential changes in the ecosystem.

CHAPTER 11. LITERATURE REVIEW UPDATE

General Conclusions

This research project was registered in the Doctoral Program in Engineering Sciences at ITESO university. The main contribution to this area is the conjunction of diverse perspectives related to Services Engineering, Computing Sciences, Systems Thinking, and Environmental and Economic Sciences, which we integrated into a nine-step methodology proposal for obtaining a simulation model to allow users to make decisions concerning circular economy (CE) strategies in their processes. This work presents the design and implementation of this methodology based on four perspectives: service-dominant logic, ecosystem services, agent-based modeling, and system dynamics.

Service-dominant logic (SDL) perspective allowed us to understand the value co-creation concept around a service ecosystem. As a CE seeks to create value from waste, SDL concepts can be applied as a guide for CE business innovation. Ecosystem services (ES) is derived from SDL as an alternate concept considering nature as another actor in a service ecosystem, providing services to people.

From these approaches, we observed the need to create a simulation model for implementing CE in the Mexican industry due to the lack of environmental impact visualization around stakeholders. The use of agent-based modeling (ABM) and system dynamics (SD) allowed us to define and design the nine-step methodology for creating a dynamic simulation model according to a specific case study.

As we found in the literature review, there is a similar framework for designing and implementing CE strategies using a multi-agent simulation. The main difference is that our methodology is focused on visualizing different scenarios in the simulator according to the strategies adopted. There is no other simulation methodology or framework for implementing CE policies in SMEs in Mexico.

This methodology was implemented in two case studies in Guadalajara, Mexico: a food bank and a confectionery factory. In both cases, a specific dynamic simulation model was developed to understand the organization's performance and to evaluate future scenarios as possible CE policies

GENERAL CONCLUSIONS

that could modify or enhance the waste used in the business processes. During the methodology development, interviews were conducted to acquire the qualitative and quantitative data needed for the simulation model. The final product delivered to the organizations' stakeholders was a dash-board where they could modify some variables and observe the state variables' performance. The model validation was conducted through a questionnaire to the stakeholders where they evaluated five characteristics of the simulation model: model purpose, model results, CE view, user interface, and simulator usage.

Regarding the CE implementation in the case studies, stakeholders were convinced that CE is essential in the actual processes. However, a financial study must evaluate the viability of its implementation. Thus, until the end of this study, CE was not implemented in the case studies' processes.

The recent literature review showed similar research regarding simulation methodologies with some differences in the case studies' implementation. Besides, the use of SDL in CE scope increases the cooperation of different actors to change the linear to circular supply chains.

In future work, the methodology can be implemented in other case studies to generalize its adoption and develop a standard simulation model for evaluating CE scenarios. This model will facilitate the CE implementation in the industries, not only to the stakeholders but also to the owners of the services related to waste and energy management, creating a service ecosystem in the metropolitan area of Guadalajara.

Conclusiones Generales

Este proyecto de investigación fue registrado en el Programa de Doctorado en Ciencias de la Ingeniería de la universidad ITESO. La principal contribución a esta área es la conjunción de diversas perspectivas relacionadas con la Ingeniería de Servicios, las Ciencias de la Computación, el Pensamiento de Sistemas y las Ciencias Ambientales y Económicas, que integramos en una propuesta metodológica de nueve pasos para obtener un modelo de simulación que permita a los usuarios tomar decisiones sobre estrategias de economía circular (EC) en sus procesos. Este trabajo presenta el diseño e implementación de esta metodología basada en cuatro perspectivas: la lógica dominante del servicio, los servicios de los ecosistemas, el modelado basado en agentes y la dinámica de sistemas.

La perspectiva de la lógica dominante del Servicio (LDS) nos permitió comprender el concepto de creación conjunta de valor en torno a un ecosistema de servicios. Dado que la EC busca crear valor a partir de los residuos, los conceptos de SDL se pueden aplicar como guía para la innovación empresarial de EC. Los servicios de los ecosistemas (ES) se derivan de LDS como un concepto alternativo que considera a la naturaleza como otro actor en un ecosistema de servicios, brindando servicios a las personas.

A partir de estos planteamientos, observamos la necesidad de crear un modelo de simulación para la implementación de EC en la industria mexicana debido a la falta de visualización del impacto ambiental en torno a los stakeholders. El uso de modelado basado en agentes (MBA) y dinámica de sistemas (DS) nos permitió definir y diseñar la metodología de nueve pasos para crear un modelo de simulación dinámica de acuerdo con un caso de estudio específico.

Como encontramos en la revisión de la literatura, existe un marco similar para diseñar e implementar estrategias de EC utilizando una simulación de múltiples agentes. La principal diferencia es que nuestra metodología se centra en visualizar diferentes escenarios en el simulador según las estrategias adoptadas. No existe otra metodología o marco de simulación para la implementación de políticas de EC en las PYMES en México.

Esta metodología se implementó en dos casos de estudio en Guadalajara, México: un banco

CONCLUSIONES GENERALES

de alimentos y una fábrica de dulces. En ambos casos, se desarrolló un modelo de simulación dinámica específico para entender el desempeño de la organización y evaluar escenarios futuros como posibles políticas de EC que pudieran modificar o potenciar los residuos utilizados en los procesos de negocio. Durante el desarrollo de la metodología, se realizaron entrevistas para adquirir los datos cualitativos y cuantitativos necesarios para el modelo de simulación. El producto final entregado a los stakeholders de las organizaciones fue un tablero donde podían modificar algunas variables y observar el desempeño de las variables de estado. La validación del modelo se realizó a través de un cuestionario a las partes interesadas donde evaluaron cinco características del modelo de simulación: propósito del modelo, resultados del modelo, vista EC, interfaz de usuario y uso del simulador.

Con respecto a la implementación de la EC en los estudios de caso, las partes interesadas estaban convencidas de que la EC es esencial en los procesos reales. Sin embargo, un estudio financiero debe evaluar la viabilidad de su implementación. Así, hasta el final de este estudio, el EC no fue implementado en los procesos de los estudios de caso.

La reciente revisión de la literatura mostró investigaciones similares con respecto a las metodologías de simulación con algunas diferencias en la implementación de los estudios de casos. Además, el uso de LDS en el ámbito de la EC aumenta la cooperación de diferentes actores para cambiar las cadenas de suministro lineales a circulares.

En trabajos futuros, la metodología puede implementarse en otros casos de estudio para generalizar su adopción y desarrollar un modelo de simulación estándar para evaluar escenarios de EC. Este modelo facilitará la implementación de la EC en las industrias, no solo a los stakeholders sino también a los titulares de los servicios relacionados con la gestión de residuos y energía, creando un ecosistema de servicios en la Zona Metropolitana de Guadalajara.

122

Appendix
A. List of Internal Research Reports

- E. L. Guevara-Rivera, R. Osorno-Hinojosa, and V. H. Zaldívar-Carrillo, "Hybrid simulation model for circular economy implementation: Literature review" Internal Report PhDEngScITESO-17-43-R, ITESO, Tlaquepaque, Mexico, Dec. 2017.
- E. L. Guevara-Rivera, R. Osorno-Hinojosa, and V. H. Zaldívar-Carrillo, "Hybrid service simulation model for circular economy implementation: Case study and methodology proposal," Internal Report PhDEngScITESO-18-15-R, ITESO, Tlaquepaque, Mexico, May 2018.
- E. L. Guevara-Rivera, R. Osorno-Hinojosa, and V. H. Zaldívar-Carrillo, "Hybrid service simulation model for circular economy implementation: Tepatitlán food bank," Internal Report PhDEngScITESO-18-48-R, ITESO, Tlaquepaque, Mexico, Dec 2018.
- E. L. Guevara-Rivera, R. Osorno-Hinojosa, and V. H. Zaldívar-Carrillo, "A simulation methodology for circular economy implementation: Agent-based modeling," Internal Report PhDEngScITESO-18-59-R, ITESO, Tlaquepaque, Mexico, Dec 2018.
- E. L. Guevara-Rivera, R. Osorno-Hinojosa, and V. H. Zaldívar-Carrillo, "A simulation methodology for circular economy implementation: Building the simulation model in NetLego," Internal Report PhDEngScITESO-19-44-R, ITESO, Tlaquepaque, Mexico, May. 2019.
- E. L. Guevara-Rivera, R. Osorno-Hinojosa, and V. H. Zaldívar-Carrillo, "A simulation methodology for circular economy implementation: Model validation," Internal Report PhDEngScITESO-19-45-R, ITESO, Tlaquepaque, Mexico, Dec. 2019.
- E. L. Guevara-Rivera, R. Osorno-Hinojosa, and V. H. Zaldívar-Carrillo, "A simulation methodology for circular economy implementation: Second case study," Internal Report PhDEngScITESO-20-25-R, ITESO, Tlaquepaque, Mexico, May. 2020.
- E. L. Guevara-Rivera, R. Osorno-Hinojosa, and V. H. Zaldívar-Carrillo, "A simulation methodology for circular economy implementation in the confectionery factory," Internal Report PhDEngScITESO-20-26-R, ITESO, Tlaquepaque, Mexico, Jul. 2020.

- E. L. Guevara-Rivera, R. Osorno-Hinojosa, and V. H. Zaldívar-Carrillo, "A simulation methodology for circular economy implementation: Circular economy measurement," Internal Report PhDEngScITESO-20-27-R, ITESO, Tlaquepaque, Mexico, Dec. 2020.
- E. L. Guevara-Rivera, R. Osorno-Hinojosa, and V. H. Zaldívar-Carrillo, "A simulation methodology for circular economy implementation: Scenario analysis," Internal Report PhDEngScITESO-21-23-R, ITESO, Tlaquepaque, Mexico, May. 2021.
- E. L. Guevara-Rivera, R. Osorno-Hinojosa, and V. H. Zaldívar-Carrillo, "A simulation methodology for circular economy implementation: Literature review update," Internal Report PhDEngScITESO-21-24-R, ITESO, Tlaquepaque, Mexico, Nov. 2021.

B. List of Publications

- E. Guevara-Rivera, R. Osorno-Hinojosa and V. H. Zaldívar-Carrillo, "Hybrid service simulation model for circular economy implementation," 9th International Congress on Environmental Modelling and Software, Fort Collins, 2018, 66. https://scholarsarchive.byu.edu/iemssconference/2018/Stream-C/66/
- E. Guevara-Rivera, R. Osorno-Hinojosa and V. H. Zaldívar-Carrillo, "A Simulation Methodology for Circular Economy Implementation," IEEE 10th International Conference on Advanced Computer Information Technologies (ACIT), Deggendorf, 2020, DOI: 10.1109/ ACIT49673.2020.9208839
- E. Guevara-Rivera, R. Osorno-Hinojosa, V. H. Zaldívar-Carrillo and H. Perez-Ortiz, "Dynamic simulation methodology for implementing circular economy: A new case study," Journal of Industrial Engineering and Management, vol. 14, no. 4, December 2021, DOI: http://dx.doi.org/10.3926/jiem.3609

Appendix B. List of Publications

Bibliography

- Javier Babí Almenar et al. "Nexus between nature-based solutions, ecosystem services and urban challenges". In: *Land use policy* 100 (2021), p. 104898.
- [2] Anylogic. AnyLogic Simulation Software. Available online: https://www.anylogic.
 com/ (accessed on 22 June 2022).
- [3] Damien Arbault et al. "Integrated earth system dynamic modeling for life cycle impact assessment of ecosystem services". In: *Science of the total environment* 472 (2014), pp. 262–272.
- [4] Stefano Armenia et al. "System Dynamics Modelling for Urban Sustainability". In: Urban Sustainability. Springer, 2021, pp. 131–173.
- [5] Weslynne Ashton. "Understanding the organization of industrial ecosystems: A social network approach". In: *Journal of Industrial Ecology* 12.1 (2008), pp. 34–51.
- [6] Sandro Azaele et al. "Dynamical evolution of ecosystems". In: *Nature* 444.7121 (2006), pp. 926–928.
- [7] Kenneth J Bagstad et al. "Spatial dynamics of ecosystem service flows: a comprehensive approach to quantifying actual services". In: *Ecosystem Services* 4 (2013), pp. 117–125.
- [8] Henrikke Baumann and Anne-Marie Tillman. "The hitch hiker's guide to LCA". In: (2004).
- [9] Claudia Marcela Betancourt Morales and Jhon Wilder Zartha Sossa. "Circular economy in Latin America: A systematic literature review". In: *Business Strategy and the Environment* 29.6 (2020), pp. 2479–2497.
- [10] Eric Bonabeau. "Agent-based modeling: Methods and techniques for simulating human systems". In: *Proceedings of the national academy of sciences* 99.suppl 3 (2002), pp. 7280–7287.
- [11] Andrei Borshchev and Alexei Filippov. "From system dynamics and discrete event to practical agent based modeling: reasons, techniques, tools". In: *Proceedings of the 22nd international conference of the system dynamics society*. Vol. 22. Citeseer. 2004, pp. 25–29.

- [12] Roelof Boumans et al. "Modeling the dynamics of the integrated earth system and the value of global ecosystem services using the GUMBO model". In: *Ecological economics* 41.3 (2002), pp. 529–560.
- [13] Roelof Boumans et al. "The Multiscale Integrated Model of Ecosystem Services (MIMES): Simulating the interactions of coupled human and natural systems". In: *Ecosystem services* 12 (2015), pp. 30–41.
- [14] Mark Brady et al. "An agent-based approach to modeling impacts of agricultural policy on land use, biodiversity and ecosystem services". In: *Landscape Ecology* 27.9 (2012), pp. 1363–1381.
- [15] Andrea Cantú, Eduardo Aguiñaga, and Carlos Scheel. "Learning from Failure and Success: The Challenges for Circular Economy Implementation in SMEs in an Emerging Economy". In: *Sustainability* 13.3 (2021), p. 1529.
- [16] Edyamira Cardozo, Yngrid Velásquez de Naime, and Carlos Rodríguez Monroy. "El concepto y la clasificación de PYME en América Latina". In: *Global Conference on Business and Finance Proceedings*. Vol. 7. 2. 2012, pp. 1657–1668.
- [17] Steve Cayzer, Percy Griffiths, and Valentina Beghetto. "Design of indicators for measuring product performance in the circular economy". In: *International Journal of Sustainable Engineering* 10.4-5 (2017), pp. 289–298.
- [18] Kai MA Chan and Terre Satterfield. "The maturation of ecosystem services: Social and policy research expands, but whither biophysically informed valuation?" In: *People and Nature* 2.4 (2020), pp. 1021–1060.
- [19] Fiona Charnley et al. "Simulation to enable a data-driven circular economy". In: Sustainability 11.12 (2019), p. 3379.
- [20] Redha Cherif and Paul Davidsson. "Software development process simulation: multi agentbased simulation versus system dynamics". In: *International Workshop on Multi-Agent Systems and Agent-Based Simulation*. Springer. 2009, pp. 73–85.
- [21] Bart Clarysse et al. "Creating value in ecosystems: Crossing the chasm between knowledge and business ecosystems". In: *Research policy* 43.7 (2014), pp. 1164–1176.

- [22] European Commission. A new Circular Economy Action Plan. Available online: https: //eur-lex.europa.eu/legal-content/EN/TXT/?qid=1583933814386&uri=COM: 2020:98:FIN (accessed on 5 October 2021).
- [23] CONEVAL. Medición de la Pobreza. Available online: https://www.coneval.org.mx/ Medicion/MP/Paginas/Pobreza_2016.aspx (accessed on 6 July 2021).
- [24] Sue ES Crawford and Elinor Ostrom. "A grammar of institutions". In: American Political Science Review 89.3 (1995), pp. 582–600.
- [25] Jonathan Cullen. "Circular economy: theoretical benchmark or perpetual motion machine?" In: (2017).
- [26] Scott A DeLoach and Jorge L Valenzuela. "An agent-environment interaction model". In: *International Workshop on Agent-Oriented Software Engineering*. Springer. 2006, pp. 1– 18.
- [27] UN Desa et al. "Transforming our world: The 2030 agenda for sustainable development". In: (2016).
- [28] Hans Dieleman and María-Concepción Martínez-Rodríguez. "Potentials and challenges for a circular economy in mexico". In: *Towards Zero Waste*. Springer, 2019, pp. 9–24.
- [29] J Emmett Duffy. "Biodiversity and ecosystem function: the consumer connection". In: *Oikos* 99.2 (2002), pp. 201–219.
- [30] EIU. Global Food Security Index. Available online: http://foodsecurityindex.eiu. com/ (accessed on 6 July 2021).
- [31] El mercado más dulce de Mexico. Available online: https://advertising-news.com/ el-mercado-mas-dulce-de-mexico/ (accessed on 21 June 2021).
- [32] Ellen MacArthur Foundation. Cities and circular economy for food. Available online: https: //www.ellenmacarthurfoundation.org/our-work/activities/cities-andcircular-economy-for-food (accessed on 21 June 2021).
- [33] Ellen MacArthur Foundation. Concept. Available online: https://www.ellenmacarthurfoundation. org/circular-economy/concept (accessed on 28 April 2021).

- [34] Thomas Elmqvist et al. "Response diversity, ecosystem change, and resilience". In: *Frontiers in Ecology and the Environment* 1.9 (2003), pp. 488–494.
- [35] Sondoss Elsawah et al. "Eight grand challenges in socio-environmental systems modeling".In: Socio-Environmental Systems Modelling 2 (2020), pp. 16226–16226.
- [36] EMAF case studies. https://www.ellenmacarthurfoundation.org/case-studies. Accessed Feb. 15, 2019.
- [37] Environment and Jalisco Government territorial development. New metropolitan model for urban solid waste management under the principle of Circular Economy. Available online: https://semadet.jalisco.gob.mx/prensa/noticia/1345 (accessed on 5 October 2021).
- [38] European Commission. SME Definition. Available online: https://ec.europa.eu/ growth/smes/sme-definition_en (accessed on 28 April 2021).
- [39] Jamie Evans and Nancy Bocken. "Circular economy toolkit". In: *Cambridge Institute for Manufacturing. http://circulareconomytoolkit. org* (2013).
- [40] Misagh Faezipour and Miad Faezipour. "System dynamics modeling for smartphone-based healthcare tools: Case study on ECG monitoring". In: *IEEE Systems Journal* (2020).
- [41] FAO. Trade Reforms and Food Security. Available online: http://www.fao.org/docrep/ 005/y4671e/y4671e06.htm#bm06.2 (accessed on 6 July 2021).
- [42] FAO. "Walking the nexus talk: assessing the water-energy-food nexus". In: *Rome* (2014).
- [43] Jacques Ferber and Olivier Gutknecht. "A meta-model for the analysis and design of organizations in multi-agent systems". In: *Proceedings international conference on multi agent systems (Cat. No. 98EX160)*. IEEE. 1998, pp. 128–135.
- [44] Filipa Ferrada and Luis M Camarinha-Matos. "A system dynamics and agent-based approach to model emotions in collaborative networks". In: *Doctoral Conference on Computing, Electrical and Industrial Systems*. Springer. 2017, pp. 29–43.
- [45] Tatiana Filatova et al. "Spatial agent-based models for socio-ecological systems: Challenges and prospects". In: *Environmental modelling & software* 45 (2013), pp. 1–7.

BIBLIOGRAPHY

- [46] Jay Wright Forrester. "Industrial dynamics". In: *Journal of the Operational Research Society* 48.10 (1997), pp. 1037–1041.
- [47] Robert A Frosch and Nicholas E Gallopoulos. "Strategies for manufacturing". In: *Scientific American* 261.3 (1989), pp. 144–153.
- [48] Andrea Genovese et al. "Sustainable supply chain management and the transition towards a circular economy: Evidence and some applications". In: *Omega* 66 (2017), pp. 344–357.
- [49] Amineh Ghorbani et al. "Model-driven agent-based simulation: Procedural semantics of a MAIA model". In: *Simulation Modelling Practice and Theory* 49 (2014), pp. 27–40.
- [50] Rick Gilsing et al. "A method for qualitative evaluation of service-dominant business models." In: *ECIS*. 2020.
- [51] Volker Grimm et al. "The ODD protocol for describing agent-based and other simulation models: A second update to improve clarity, replication, and structural realism". In: *Journal of Artificial Societies and Social Simulation* 23.2 (2020).
- [52] Volker Grimm et al. "The ODD protocol: a review and first update". In: *Ecological mod-elling* 221.23 (2010), pp. 2760–2768.
- [53] CN Guerrero, Philipp Schwarz, and Jill H Slinger. "A recent overview of the integration of system dynamics and agent-based modelling and simulation". In: *System Dynamics Conference*. 2016, pp. 1–13.
- [54] E. Guevara-Rivera, R. Osorno-Hinojosa, and V. Zaldívar-Carrillo. "Hybrid service simulation model for circular economy implementation". In: *Int. Congress on Environmental Modelling and Software*. Forth Collins CO, June 2018, pp. 1–11.
- [55] Edna Guevara-Rivera, Roberto Osorno-Hinojosa, and Victor Hugo Zaldivar-Carrillo. "A simulation methodology for circular economy implementation". In: In Proceedings of the 10th International Conference on Advanced Computer Information Technologies (ACIT). Deggendorf, Germany, September 2020.

- [56] Edna Guevara-Rivera, Roberto Osorno-Hinojosa, and Victor Hugo Zaldivar-Carrillo. "Hybrid Service Simulation Model for Circular Economy Implementation". In: *In Proceedings* of the 9th International Congress on Environmental Modelling and Software. Forth Collins, USA, June 2018.
- [57] Willi Haas et al. "How circular is the global economy?: An assessment of material flows, waste production, and recycling in the European Union and the world in 2005". In: *Journal* of industrial ecology 19.5 (2015), pp. 765–777.
- [58] Susan TL Harrison et al. "Value Recovery From Solid Confectionery Waste". In: (2019).
- [59] Scott Heckbert, Tim Baynes, and Andrew Reeson. "Agent-based modeling in ecological economics". In: *Annals of the New York Academy of Sciences* 1185.1 (2010), pp. 39–53.
- [60] Reza Hesan. "Multi-methodology Modeling to Support Policy Analysis in Socio-technical Systems". In: (2016).
- [61] Jani Holopainen et al. "Changing institutions and consumer-driven development of forest products and services". In: *Dissertationes Forestales* (2016).
- [62] Eleni Iacovidou et al. "Metrics for optimising the multi-dimensional value of resources recovered from waste in a circular economy: A critical review". In: *Journal of Cleaner Production* 166 (2017), pp. 910–938.
- [63] ITESO. Misión del ITESO. Available online: https://www.iteso.mx/web/general/ detalle?group id=30156 (accessed on 6 July 2021).
- [64] Agnieszka Janik and Adam Ryszko. "Towards measuring circularity at product level–Methodology and application of material circularity indicator". In: *7th Carpathian Logistics Congress-CLC*. 2017.
- [65] Alexandra Jurgilevich et al. "Transition towards circular economy in the food system". In: *Sustainability* 8.1 (2016), p. 69.
- [66] Roope Oskari Kaaronen and Nikita Strelkovskii. "Cultural evolution of sustainable behaviors: Pro-environmental tipping points in an agent-based model". In: *One Earth* 2.1 (2020), pp. 85–97.

- [67] Yuliya Kalmykova, Madumita Sadagopan, and Leonardo Rosado. "Circular economy–From review of theories and practices to development of implementation tools". In: *Resources, Conservation and Recycling* 135 (2018), pp. 190–201.
- [68] Dean C Karnopp, Donald L Margolis, and Ronald C Rosenberg. *System dynamics: modeling, simulation, and control of mechatronic systems.* John Wiley & Sons, 2012.
- [69] Jack PC Kleijnen. "Validation of models: statistical techniques and data availability". In: Proceedings of the 31st conference on Winter simulation: Simulation—a bridge to the future-Volume 1. 1999, pp. 647–654.
- [70] Jouni Korhonen, Antero Honkasalo, and Jyri Seppälä. "Circular economy: the concept and its limitations". In: *Ecological economics* 143 (2018), pp. 37–46.
- [71] Arnis Lektauers, Jēkabs Trušiņš, and Inese Trušiņa. "Combined multi-scale system dynamics and agent-based framework for sustainable community modelling". In: (2011).
- [72] Teresa Shamah Levy, Maritza Alejandra Amaya Castellanos, Lucía Cuevas Nasu, et al."Desnutrición y obesidad: doble carga en México". In: (2015).
- [73] Michael Lieder et al. "Towards circular economy implementation in manufacturing systems using a multi-method simulation approach to link design and business strategy". In: *The International Journal of Advanced Manufacturing Technology* 93.5 (2017), pp. 1953–1970.
- [74] Jakob Lundberg and Fredrik Moberg. "Mobile link organisms and ecosystem functioning: implications for ecosystem resilience and management". In: *Ecosystems* 6.1 (2003), pp. 0087–0098.
- [75] Miguel Lurgi and Francesc Estanyol. "MADBE: A Multi-Agent Digital Business Ecosystem". In: 4th IEEE International Conference on Digital Ecosystems and Technologies. IEEE. 2010, pp. 262–267.
- [76] Robert F Lusch. "Reframing supply chain management: a service-dominant logic perspective". In: *Journal of supply chain management* 47.1 (2011), pp. 14–18.
- [77] Robert F Lusch and Nicholas SP Tay. "Agent-based modeling: Gaining insight into firm and industry performance". In: *Assessing marketing strategy performance* (2004), pp. 213–227.

- [78] Zheng Ma. "Business ecosystem modeling-the hybrid of system modeling and ecological modeling: an application of the smart grid". In: *Energy Informatics* 2.1 (2019), pp. 1–24.
- [79] Charles M Macal and Michael J North. "Tutorial on agent-based modeling and simulation".In: *Proceedings of the Winter Simulation Conference*, 2005. IEEE. 2005, 14–pp.
- [80] E MacArthur. "Circularity indicators: An approach to measuring circularity". In: *Methodology* (2015).
- [81] Ellen MacArthur et al. *Ellen MacArthur Foundation Publications*. Available online: https://ellenmacarthurfoundation.org/publications (accessed on 5 October 2021).
- [82] Ellen MacArthur et al. "Towards the circular economy". In: *Journal of Industrial Ecology* 2 (2013), pp. 23–44.
- [83] Maken. Maken, Sustainability Agency. Available online: https://www.maken.mx/ (accessed on 11 November 2021).
- [84] Simone Manfredi et al. "Improving sustainability and circularity of European food waste management with a life cycle approach". In: *Publications Office of the European Union* (2015).
- [85] Romina Martin and Maja Schlüter. "Combining system dynamics and agent-based modeling to analyze social-ecological interactions—an example from modeling restoration of a shallow lake". In: *Frontiers in Environmental Science* 3 (2015), p. 66.
- [86] Brent D Matthies et al. "An ecosystem service-dominant logic?-integrating the ecosystem service approach and the service-dominant logic". In: *Journal of cleaner production* 124 (2016), pp. 51–64.
- [87] Luis Mazón Arevalo et al. "Análisis de punto de equilibrio en la toma de decisiones de un negocio: caso Grand Bazar Riobamba–Ecuador". In: *Revista de estrategias del desarrollo empresarial* (2017), pp. 14–24.
- [88] Jamal Hussain Miah. "From Factory to Supply Chain: Reducing Environmental Impacts of Confectionery Manufacturing using Heat Integration and Life Cycle Assessment". PhD thesis. University of Surrey (United Kingdom), 2018.

BIBLIOGRAPHY

- [89] L Scott Mills, Michael E Soulé, and Daniel F Doak. "The keystone-species concept in ecology and conservation". In: *BioScience* 43.4 (1993), pp. 219–224.
- [90] Dave Murray-Rust et al. "Conceptualising the analysis of socio-ecological systems through ecosystem services and agent-based modelling". In: *Journal of Land Use Science* 6.2-3 (2011), pp. 83–99.
- [91] Claudio Nägeli et al. "Towards agent-based building stock modeling: Bottom-up modeling of long-term stock dynamics affecting the energy and climate impact of building stocks".
 In: *Energy and Buildings* 211 (2020), p. 109763.
- [92] United Nations. "Transforming our world: the 2030 agenda for sustainable development". In: *General Assembly* (2015).
- [93] OECD. "Material productivity". In: (2019). DOI: https://doi.org/https://doi.org/ 10.1787/dae52b45-en. URL: https://www.oecd-ilibrary.org/content/data/ dae52b45-en.
- [94] Oficina Comunicación Institucional. Investigación con Pertinencia Social. Available online: https://cruce.iteso.mx/investigacion-con-pertinencia-social/ (accessed on 6 July 2021).
- [95] Effie Papargyropoulou et al. "The food waste hierarchy as a framework for the management of food surplus and food waste". In: *Journal of Cleaner Production* 76 (2014), pp. 106–115.
- [96] Mirva Peltoniemi and Elisa Vuori. "Business ecosystem as the new approach to complex adaptive business environments". In: *Proceedings of eBusiness research forum*. Vol. 2. 22. 2004, pp. 267–281.
- [97] Gunnar Petter et al. "Agent-based modeling of the effects of forest dynamics, selective logging, and fragment size on epiphyte communities". In: *Ecology and evolution* 11.6 (2021), pp. 2937–2951.
- [98] Marina PP Pieroni, Tim C McAloone, and Daniela CA Pigosso. "Business model innovation for circular economy and sustainability: A review of approaches". In: *Journal of cleaner production* 215 (2019), pp. 198–216.

- [99] Ales Polic and Karel Jezernik. "Closed-loop matrix based model of discrete event systems for machine logic control design". In: *IEEE Transactions on Industrial Informatics* 1.1 (2005), pp. 39–46.
- [100] Michael E Porter. "The value chain and competitive advantage". In: *Understanding business processes* (2001), pp. 50–66.
- [101] Michael E Porter and Mark R Kramer. "Strategy and Society, the link between competitive advantage and corporate social responsibility". In: *Harvard business review* 84.12 (2006), pp. 78–92.
- [102] José Potting et al. *Circular economy: measuring innovation in the product chain*. 2544. PBL Publishers, 2017.
- [103] Robin G Qiu. "Computational thinking of service systems: Dynamics and adaptiveness modeling". In: *Service Science* 1.1 (2009), pp. 42–55.
- [104] Steven F Railsback and Volker Grimm. *Agent-based and individual-based modeling: a practical introduction.* Princeton university press, 2019.
- [105] Chathura Rajapakse and Takao Terano. "An agent-based implementation of service system interactions based on the ISPAR model". In: Agent-Based Approaches in Economic and Social Complex Systems VIII. Springer, 2015, pp. 177–188.
- [106] Chathura Rajapakse and Takao Terano. "An agent-based model to study the evolution of service systems through the service life cycle". In: *Int. J. Energy, Inf. Commun* 4.5 (2013).
- [107] Beatrice Re, Giovanna Magnani, and Antonella Zucchella. "The Future of Sustainability: Value Co-creation Processes in the Circular Economy". In: *The Palgrave Handbook of Corporate Sustainability in the Digital Era*. Springer, 2021, pp. 503–527.
- [108] Ke Rong et al. "Understanding business ecosystem using a 6C framework in Internet-of-Things-based sectors". In: *International Journal of Production Economics* 159 (2015), pp. 41– 55.
- [109] MDA Rounsevell, DT Robinson, and Dave Murray-Rust. "From actors to agents in socioecological systems models". In: *Philosophical Transactions of the Royal Society B: Biological Sciences* 367.1586 (2012), pp. 259–269.

- [110] Bonnie Rubenstein-Montano et al. "A systems thinking framework for knowledge management". In: *Decision support systems* 31.1 (2001), pp. 5–16.
- [111] Andrew P Sage and William B Rouse. *Handbook of systems engineering and management*. John Wiley & Sons, 2014.
- [112] Michael Saidani et al. "A taxonomy of circular economy indicators". In: *Journal of Cleaner Production* 207 (2019), pp. 542–559.
- [113] Michael Saidani et al. "Hybrid top-down and bottom-up framework to measure products" circularity performance". In: 2017.
- [114] Michael Saidani et al. "Testing the robustness of circularity indicators: empirical insights from workshops on an industrial product". In: *Proceedings of the Design Society: International Conference on Engineering Design*. Vol. 1. 1. Cambridge University Press. 2019, pp. 3401–3410.
- [115] Furkan Sariatli. "Linear economy versus circular economy: A comparative and analyzer study for optimization of economy for sustainability". In: *Visegrad Journal on Bioeconomy* and Sustainable Development 6.1 (2017), pp. 31–34.
- [116] Nadine Schieritz and Peter M Milling. "Modeling the forest or modeling the trees". In: Proceedings of the 21st International Conference of the System Dynamics Society. Citeseer. 2003, pp. 20–24.
- [117] Markus Schwaninger. "System dynamics in the evolution of the systems approach". In: System Dynamics: Theory and Applications (2020), pp. 21–39.
- [118] Suvi Seikkula. ""Show Me the Value": Business Benefits of Co-creation and How to Effectively Communicate Their Value to SMEs Which Are Developing Circular Economy Business Models". In: (2020).
- [119] Hillary Sillitto et al. "Defining "system": A comprehensive approach". In: *INCOSE International Symposium*. Vol. 27. 1. Wiley Online Library. 2017, pp. 170–186.
- [120] John D Sterman. Business dynamics: systems thinking and modeling for a complex world.HD30. 2 \$7835 2000. 2000.

- [121] Zhanli Sun and Daniel Müller. "A framework for modeling payments for ecosystem services with agent-based models, Bayesian belief networks and opinion dynamics models". In: *Environmental modelling & software* 45 (2013), pp. 15–28.
- [122] Chris Swinerd and Ken R McNaught. "Design classes for hybrid simulations involving agent-based and system dynamics models". In: *Simulation Modelling Practice and Theory* 25 (2012), pp. 118–133.
- [123] System Dynamics Society. What is System Dynamics? Available online: https://www. systemdynamics.org/what-is-sd (accessed on 28 April 2021).
- [124] Fabian Takacs, Karolin Frankenberger, and Richard Stechow. "Circular Ecosystems: Business Model Innovation for the Circular Economy". In: (2020).
- [125] Arthur G Tansley. "The use and abuse of vegetational concepts and terms". In: *Ecology* 16.3 (1935), pp. 284–307.
- [126] Maarika Teose et al. "Embedding system dynamics in agent based models for complex adaptive systems". In: *Twenty-Second International Joint Conference on Artificial Intelligence*. 2011.
- [127] Takao Terano and Chathura Rajapakse. "An agent-based model to study customers' engagement with brands from a service-dominant logic perspective". In: Social Simulation Conference. 2014.
- [128] David Tilman et al. "Agricultural sustainability and intensive production practices". In: *Nature* 418.6898 (2002), pp. 671–677.
- [129] Trade reforms and food security. http://www.fao.org/docrep/005/y4671e/ y4671e06.htm#bm06. Accessed March 23, 2020.
- [130] UNESCO. The United Nations world water development report 2020: water and climate change. Available online: https://unesdoc.unesco.org/ark:/48223/pf0000372985 (accessed on 21 June 2021).
- [131] Stephen L Vargo. "Beyond Circularity—A Service-dominant (SD) Logic Perspective". In: *Circular Economy and Sustainability* (2021), pp. 1–4.

- [132] Stephen L Vargo, Kaisa Koskela-Huotari, and Josina Vink. "Service-dominant logic: foundations and applications". In: *The Routledge handbook of service research insights and ideas* (2020), pp. 3–23.
- [133] Stephen L Vargo and Robert F Lusch. "Evolving to a new dominant logic for marketing". In: *Journal of marketing* 68.1 (2004), pp. 1–17.
- [134] Stephen L Vargo and Robert F Lusch. "Institutions and axioms: an extension and update of service-dominant logic". In: *Journal of the Academy of marketing Science* 44.1 (2016), pp. 5–23.
- [135] Stephen L Vargo and Robert F Lusch. "Service-dominant logic 2025". In: International Journal of Research in Marketing 34.1 (2017), pp. 46–67.
- [136] Stephen L Vargo and Robert F Lusch. "Service-dominant logic: continuing the evolution".In: *Journal of the Academy of marketing Science* 36.1 (2008), pp. 1–10.
- [137] WBCSD. Circular Transition Indicators. Available online: https://www.wbcsd.org/ Programs/Circular-Economy/Factor-10/Metrics-Measurement/Circulartransition-indicators (accessed on 28 April 2021).
- [138] Ken Webster. *The circular economy: A wealth of flows*. Ellen MacArthur Foundation Publishing, 2017.
- [139] Uri Wilensky and William Rand. *An introduction to agent-based modeling: modeling natural, social, and engineered complex systems with NetLogo.* MIT Press, 2015.
- [140] Michael Wooldridge. An introduction to multiagent systems. John Wiley & Sons, 2009.
- [141] Devrim Murat Yazan and Luca Fraccascia. "Sustainable operations of industrial symbiosis: an enterprise input-output model integrated by agent-based simulation". In: *International Journal of Production Research* 58.2 (2020), pp. 392–414.
- [142] Christoph Zott and Raphael Amit. "Business model design: An activity system perspective".In: *Long range planning* 43.2-3 (2010), pp. 216–226.